



Available online at www.sciencedirect.com



Comput. Methods Appl. Mech. Engrg. 392 (2022) 114680

Computer methods in applied mechanics and engineering

www.elsevier.com/locate/cma

# A New Updated Reference Lagrangian Smooth Particle Hydrodynamics algorithm for isothermal elasticity and elasto-plasticity

Paulo R. Refachinho de Campos<sup>a,c</sup>, Antonio J. Gil<sup>a,\*</sup>, Chun Hean Lee<sup>b,\*</sup>, Matteo Giacomini<sup>c,d</sup>, Javier Bonet<sup>e</sup>

<sup>a</sup> Zienkiewicz Centre for Computational Engineering, Faculty of Science and Engineering, Swansea University, Bay Campus, SA1 8EN, United

Kingdom

<sup>b</sup> Glasgow Computational Engineering Centre, James Watt School of Engineering, University of Glasgow, United Kingdom <sup>c</sup> Laboratori de Càlcul Numèric (LaCàN), ETS de Ingenieros de Caminos, Canales y Puertos, Universitat Politècnica de Catalunya, Barcelona, Spain

<sup>d</sup> International Centre for Numerical Methods in Engineering (CIMNE), Barcelona, Spain <sup>e</sup> University of Greenwich, London, SE10 9LS, United Kingdom

Received 21 September 2021; received in revised form 18 January 2022; accepted 22 January 2022 Available online xxxx

# Abstract

This paper presents a new Updated Reference Lagrangian Smooth Particle Hydrodynamics (SPH) algorithm for the analysis of large deformation isothermal elasticity and elasto-plasticity. Taking as point of departure a Total Lagrangian setting and considering as referential configuration an intermediate configuration of the deformation process, the equation of conservation of linear momentum and three geometric conservation laws (for the deformation gradient, its cofactor and its determinant) are re-written leading to a very generic system of first order conservation laws. The key feature of the formulation is a suitable multiplicative decomposition of the conservation variables, leading to a very simple final set of equations with striking similarities to the conventional Total Lagrangian system albeit re-written in terms of alternative Referential Updated conservation variables which are evolved in time. Taking advantage of this new Updated Reference Lagrangian formalism, a second order entropy-stable SPH upwinding stabilisation method will be introduced. With respect to previous publications by the group, a new three-stage Runge–Kutta time integration method is implemented in order to increase the CFL stability restriction. Finally, and to demonstrate the robustness and applicability of the methodology, a wide spectrum of challenging problems. To facilitate its ease of implementation, the paper explores the use of a series of novel expressions for the evaluation of kernels and the gradients of kernels to provide the SPH user the freedom to choose amongst various options, all leading to equally convincingly robust results.

© 2022 Published by Elsevier B.V.

Keywords: SPH; Updated reference Lagrangian; Conservation laws; Upwinding; Fast dynamics

\* Corresponding authors. *E-mail addresses:* a.j.gil@swansea.ac.uk (A.J. Gil), chunhean.lee@glasgow.ac.uk (C.H. Lee).

https://doi.org/10.1016/j.cma.2022.114680 0045-7825/© 2022 Published by Elsevier B.V.

## 1. Introduction

Numerical methods that do not require connection between nodes or the creation of sub-regions (elements) in the process of discretisation comprise a large family of methods called meshless (or meshfree) methods, see e.g. Belytschko et al. [1], Huerta et al. [2] and Chen et al. [3] for in-depth studies. The main objective of such techniques is to eliminate some of the limitations related to mesh dependence of conventional computational methods. Since the approximations are constructed in terms of nodes (particles) and not based on elements (their distortion may degrade accuracy), meshfree methods can address a range of problems much wider than mesh-based methods. Moreover, meshfree methods [4] have the potential of circumventing some difficulties usually faced in the pre-processing stage, i.e., meshfree methods eliminate the standard mesh generation process, making the integration between Computer Aided Engineering (CAE) and Computer Aided Design (CAD) easier [5].

One of the earliest meshfree methods is the Smooth Particle Hydrodynamics (SPH), first presented in 1977 by Gingold et al. [6] and Lucy [7]. Libersky et al. were amongst the first to use the SPH method in the context of strength of materials [8] and dynamic solid mechanics [9]. The SPH method gives numerical solutions to initial-boundary value problems defined by conservation laws of continuum mechanics combined with constitutive relations for the materials involved. It is important to emphasise that SPH is not based on discrete particles colliding with each other or presenting cohesive-like behaviour. Instead, SPH is a fully Lagrangian modelling scheme that permits the discretisation of continuum partial differential equations through the approximation of properties at a set of points distributed over the domain, without the need of defining a spatial mesh. The Lagrangian nature of the method and its ability to keep accurate history of events, associated with its meshless characteristics, represent the main advantages of SPH. A massive number of publications regarding the SPH method is available in the literature. The works by Monaghan [10], Swegle et al. [11] and Vignjevic et al. [12] and references therein provide comprehensive reviews of the topic.

The classical Eulerian and Updated Lagrangian displacement-based SPH methods are known to suffer from a number of well-known drawbacks, such as numerical instabilities in the form of spurious zero-energy modes and tensile instability (the latter only in the case of Eulerian SPH) [13-26], lack of consistency [18,20,27,28], loss of conservation [24,29,30] and reduced order of convergence for derived variables such as stresses and strains [24]. Distinct research groups [31–44] have come up with various solutions for the different issues faced in SPH. Some interesting work has also been reported in [45,46] where a hourglass control based stabilisation algorithm is employed for the description of elasto- and visco-plastic continuum. It is however still not yet clear how to introduce appropriate numerical viscosity through the classical Coleman-Noll procedure in order to ensure the production of non-negative numerical entropy. Additionally, the kernel function as well as the gradient of its kernel used in [45,46] are evaluated based on standard updated isotropic (spherical) kernels, which may not be optimal in the presence of strong anisotropic changes in volume. In the recent work by Lee et al. [26], a Riemann solver based stabilisation strategy for a mixed-based SPH framework for large strain explicit solid dynamics was presented. The framework distinguishes from standard SPH techniques due to the fact that the conservation of the linear momentum p is solved along with the conservation equations for the deformation gradient F and its minors. Conservation laws of geometric quantities such as the deformation gradient F, the volume map (Jacobian) J and the area map (cofactor) **H** are introduced aiming at the establishment of a mixed set of conservation laws. In previous works, but in different contexts, Lee et al. [47,48] proposed other methodologies which had as unknowns the linear momentum, the deformation gradient and the total energy. With both the linear momentum and the deformation gradient tensor being primary variables of the problem, stresses converge at the same rate as the velocities and displacements. Moreover, the new formulation was shown to be efficient in nearly incompressible and bending dominated scenarios. However, in case of extreme deformations in the incompressible limit, the  $\{p, F\}$  formulation lacks robustness. In [49], in the context of Finite Element Method (FEM), Gil et al. enhanced the formulation for nearly and truly incompressible deformations with the novelty of introducing a conservation law for the Jacobian J of the deformation gradient, providing extra flexibility to the scheme. In [50], a new geometric conservation law for the co-factor H of the deformation gradient was also added to the framework, leading to an enhanced mixed-formulation, very amenable for constitutive laws strongly dependent on the cofactor of the deformation. The development of this new conservation equation was possible due to the simplification introduced by the use of a tensor cross product operation, presented for the first time in the context of solid mechanics in [51] and further explained in [52].

The evolution of the above mentioned works resulted in a mixed methodology in the form of a system of Total Lagrangian first order conservation laws  $\{p, F, H, J\}$ . Following these developments, a new stabilised total

Lagrangian SPH methodology was introduced in [26], aiming at the removal of spurious zero-energy modes and improved order of convergence of variables such as stresses and strains. Moreover, in this approach, linear consistency is fulfilled by performing kernel and gradient corrections, as proposed in [28]. Global conservation of angular momentum is strictly ensured with the introduction of a monolithic angular momentum projection algorithm, first presented in [24]. The stabilisation methodology presented in [26] does not require user-defined artificial stabilisation parameters and is based on physical pressure and shear wave speeds.

Even though the combination of all these developments has greatly improved the current state of SPH methods in fast solid dynamics, there is still the lack of a robust updated Lagrangian SPH framework, which is necessary when topological changes (i.e. fracture, separation) in the geometry need to be taken into account along the solution. With this idea in mind, the present work puts forward a new stabilised Updated Reference Lagrangian SPH framework, aimed to be a robust alternative tool for computer simulations in fast solid dynamics problems, prepared for the future endeavour of handling problems with topological changes. Specifically, and by adopting as referential configuration an intermediate configuration of the deformation process, the equation of conservation of linear momentum and three geometric conservation laws (for the deformation gradient, its cofactor and its determinant) are re-written leading to a very generic (incremental) system of first order conservation laws. One attractive feature of the methodology is its possibility to degenerate into either Total or purely Updated Lagrangian formulations. The actual use of the methodology in the context of fracture mechanics [53] is the next step of this work, beyond the current scope, along with the consideration of thermal effects induced by strong thermo-mechanical coupling due to shocks.

The paper is broken-down into the following sections. Section 2 starts by presenting the Total Lagrangian Formulation of the conservation laws to be solved, encompassing the linear momentum and three geometric conservation laws. The section continues deriving the new Updated Reference Lagrangian set of conservation laws in terms of a new set of alternative conservation variables and introducing their appropriate (entropy) work conjugates. Section 3 presents the variational statements of the problem. Section 4 presents the Smooth Particle Hydrodynamics discretisation scheme. Special attention is paid to two aspects: first, the Riemann based (upwinding) numerical dissipation employed; second, the introduction of four different options for the evaluation of kernels and gradients of kernels needed for SPH implementation. Section 5 briefly presents the three-stage explicit Runge–Kutta scheme used to evolve the semi-discrete equations in time, improving the numerical CFL restriction previously published in the literature. Section 6 presents a simple projection technique to ensure strict total conservation of angular momentum. Section 7 includes the algorithmic flowchart of the resulting numerical scheme and how to degenerate into their Total and purely Updated Lagrangian algorithmic counterparts. Section 8 presents a set of numerical examples to assess the convergence, conservation and stability of the computational framework, with detailed comparison with existing implementations available in the literature. Section 9 presents some concluding remarks.

## 2. Reversible elastodynamics

## 2.1. A summary of the total Lagrangian formalism

Consider the three dimensional deformation of an isothermal body of material density  $\rho_R$  moving from its initial undeformed configuration  $\Omega_V$ , with boundary  $\partial \Omega_V$  defined by an outward unit normal N, to a current deformed configuration  $\Omega_v$  at time t, with boundary  $\partial \Omega_v$  defined by an outward unit normal n (see Fig. 1). The time dependent motion  $\phi(X, t)$  of the body can be described by a system of first order Total Lagrangian conservation laws [24–26,54,55], summarised as

$$\frac{\partial \boldsymbol{\mathcal{U}}}{\partial t} + \sum_{I=1}^{3} \frac{\partial \boldsymbol{\mathcal{F}}_{I}}{\partial X_{I}} = \boldsymbol{\mathcal{S}} \qquad \text{in} \qquad \boldsymbol{\Omega}_{V}.$$
(1)

Here,  $\mathcal{U}$  is the vector of conservation variables,  $\mathcal{F}_{I}$  is the flux column vector in *I*th material direction and  $\mathcal{S}$  is the source term, described as

$$\mathcal{U} = \begin{bmatrix} p \\ F \\ H \\ J \end{bmatrix}, \qquad \mathcal{F}_I = -\begin{bmatrix} PE_I \\ v \otimes E_I \\ F \times (v \otimes E_I) \\ H : (v \otimes E_I) \end{bmatrix}, \qquad \mathcal{S} = \begin{bmatrix} f_R \\ 0 \\ 0 \\ 0 \end{bmatrix}.$$
(2)



Fig. 1. Deformation of a solid body.

In above system of equations,  $p = \rho_R v$  is the linear momentum per unit material volume, v represents the velocity field,  $f_R$  is the body force per unit material volume,  $\{F, H, J\}^1$  represent the deformation gradient tensor, its co-factor and its Jacobian, P represents the first Piola–Kirchhoff stress tensor and the material identity tensor I is expressed as  $I = \sum_{I=1}^{3} E_I \otimes E_I$  with the Cartesian material coordinate basis being defined as

$$\boldsymbol{E}_{1} = \begin{bmatrix} 1\\0\\0 \end{bmatrix}; \quad \boldsymbol{E}_{2} = \begin{bmatrix} 0\\1\\0 \end{bmatrix}; \quad \boldsymbol{E}_{3} = \begin{bmatrix} 0\\0\\1 \end{bmatrix}.$$
(3)

DIV and CURL represent the material divergence and curl operators carried out with respect to the material configuration, and the symbol  $\times$  represents the tensor cross product between vectors and/or second order tensors, see [52].

As the system of conservation laws presented above has more equations than needed, suitable compatibility relationships (also known as involutions [47,48,56,57]) are necessary, namely

$$CURL F = 0; \qquad DIV H = 0. \tag{4}$$

For closure of the system above, a suitable constitutive relationship for the stresses P compliant with the principle of objectivity (frame invariance) and the second law of Thermodynamics (Coleman–Noll procedure) is presented in Remark 1. Finally, for the complete definition of the initial boundary value problem, both initial and boundary (essential and natural) conditions must also be specified as appropriate.

**Remark 1.** In this work, and without loss of generality, a Mooney–Rivlin hyperelastic constitutive model has been considered, where the strain energy density is defined as a convex multi-variable function W of the triplet of deformation measures {F, H, J} [50,58,59] as

$$W(F, H, J) = \zeta J^{-2/3} (F : F) + \xi J^{-2} (H : H)^{3/2} - 3 \left(\zeta + \sqrt{3}\xi\right) + \frac{\kappa}{2} (J - 1)^2,$$
(5)

where  $\zeta$ ,  $\xi$  and  $\kappa$  (bulk modulus) are positive material parameters. By comparison of the tangent elasticity operator at the initial configuration with that of classical linear elasticity, appropriate values for the material parameters  $\zeta$ and  $\xi$  can be defined in terms of the shear modulus  $\mu$ , that is,  $2\zeta + 3\sqrt{3}\xi = \mu$  [58]. Following Refs. [24–26], the first Piola Kirchhoff stress tensor can be expressed as

$$\boldsymbol{P} = \boldsymbol{\Sigma}_{\boldsymbol{F}} + \boldsymbol{\Sigma}_{\boldsymbol{H}} \times \boldsymbol{F} + \boldsymbol{\Sigma}_{\boldsymbol{J}} \boldsymbol{H},\tag{6}$$

<sup>1</sup> Classically, the definitions for the triplet of deformation measures are  $F = \nabla_0 \phi$ ,  $H = \frac{1}{2}F \times F$  and  $J = \det F$ , respectively.

where the conjugate stresses  $\{\Sigma_F, \Sigma_H, \Sigma_J\}$  are defined by

$$\boldsymbol{\Sigma}_{\boldsymbol{F}} = \frac{\partial W}{\partial \boldsymbol{F}} = 2\zeta J^{-2/3} \boldsymbol{F}; \qquad \boldsymbol{\Sigma}_{\boldsymbol{H}} = \frac{\partial W}{\partial \boldsymbol{H}} = 3\xi J^{-2} (\boldsymbol{H} : \boldsymbol{H})^{1/2} \boldsymbol{H}$$
(7)

and

$$\Sigma_J = \frac{\partial W}{\partial J} = -\frac{2}{3} \zeta J^{-5/3}(\boldsymbol{F} : \boldsymbol{F}) - 2\xi J^{-3}(\boldsymbol{H} : \boldsymbol{H})^{3/2} + \kappa(J-1).$$
(8)

#### 2.2. Updated reference Lagrangian formalism

Taking inspiration from the work presented in Ref. [23,45,46], an equivalent system of Updated Reference Lagrangian conservation equations can be written in terms of a new configuration, additional to the well established material (initial) and spatial (current) configurations. This new configuration is defined as an intermediate or incremental configuration through which the continuum has passed (convected) during the deformation process at a previous time instant or load increment (refer to the so-called *"Fixed" in time* configuration in Fig. 1). This extra configuration can be adopted as a new reference configuration, leading to what we will refer to as a Updated Reference Lagrangian formulation. As we will see, this additional referential (intrinsic) configuration leads to an enhanced formulation, capable of degenerating into the classical Total Lagrangian or Updated Lagrangian formalisms.

By doing this, the triplet of deformation measures  $\{F, H, J\}$  (from material to spatial configurations) can be obtained via a multiplicative decomposition as

$$F = f F_{\chi}; \qquad H = h H_{\chi}; \qquad J = j J_{\chi}, \tag{9}$$

where  $\{f, h, j\}$  represents the set of (incremental) deformations measured from the reference to the spatial configuration, whereas  $\{F_{\chi}, H_{\chi}, J_{\chi}\}$  denotes the set of (known) deformations measured from the material to the reference configuration.

In order to derive the Updated Reference Lagrangian formalism, we can utilise appropriate push forward transformations for vector and flux quantities [60], that is, multiplying material vector quantities  $\{\mathcal{U}, \mathcal{S}\}$  by  $J_{\chi}^{-1}$  and post-multiplying the Total Lagrangian flux  $\mathcal{F}$  by  $H_{\chi}^{-1}$ , making use of the Piola identity

$$\sum_{I=1}^{3} \frac{\partial \mathcal{F}_{I}}{\partial X_{I}} = J_{\chi} \sum_{i=1}^{3} \frac{\partial \left(\mathcal{F} H_{\chi}^{-1}\right)_{i}}{\partial \chi_{i}},\tag{10}$$

Eq. (1) thus becomes

•

$$J_{\chi}^{-1} \frac{\partial \mathcal{U}}{\partial t} + \sum_{i=1}^{3} \frac{\partial \left(\mathcal{F} \boldsymbol{H}_{\chi}^{-1}\right)_{i}}{\partial \chi_{i}} = J_{\chi}^{-1} \boldsymbol{\mathcal{S}}; \quad \text{in} \quad \Omega_{\chi},$$
(11)

which can then be particularised for the individual components of  $\mathcal{U}$ , yielding

$$J_{\chi}^{-1} \frac{\partial \boldsymbol{p}}{\partial t} - \operatorname{div}_{\chi} \left( \boldsymbol{P} \boldsymbol{H}_{\chi}^{-1} \right) = J_{\chi}^{-1} \boldsymbol{f}_{R}, \tag{12}$$

$$J_{\chi}^{-1} \frac{\partial F}{\partial t} - \operatorname{div}_{\chi} \left( \boldsymbol{v} \otimes \boldsymbol{H}_{\chi}^{-1} \right) = \boldsymbol{0}, \tag{13}$$

$$J_{\chi}^{-1} \frac{\partial \boldsymbol{H}}{\partial t} - \operatorname{div}_{\chi} \left( \boldsymbol{F} \times \left( \boldsymbol{v} \otimes \boldsymbol{H}_{\chi}^{-1} \right) \right) = \boldsymbol{0}, \tag{14}$$

$$J_{\chi}^{-1} \frac{\partial J}{\partial t} - \operatorname{div}_{\chi} \left( \boldsymbol{v} \cdot \left( \boldsymbol{H} \boldsymbol{H}_{\chi}^{-1} \right) \right) = 0.$$
<sup>(15)</sup>

Additionally, we introduce the following intermediate variables defined as

$$\boldsymbol{p}_{\boldsymbol{\chi}} = \rho_{\boldsymbol{\chi}} \boldsymbol{v}; \qquad \rho_{\boldsymbol{\chi}} = J_{\boldsymbol{\chi}}^{-1} \rho_{\boldsymbol{R}}; \qquad \boldsymbol{\sigma}_{\boldsymbol{\chi}} = \boldsymbol{P} \boldsymbol{H}_{\boldsymbol{\chi}}^{-1}; \qquad \boldsymbol{f}_{\boldsymbol{\chi}} = J_{\boldsymbol{\chi}}^{-1} \boldsymbol{f}_{\boldsymbol{R}}. \tag{16}$$

With these, and with use of (9), above system ((12)-(15)) after some algebraic manipulation reduces to

$$\frac{\partial \boldsymbol{p}_{\chi}}{\partial t} = \operatorname{div}_{\chi} \boldsymbol{\sigma}_{\chi} + \boldsymbol{f}_{\chi}, \tag{17}$$

P.R.R. de Campos, A.J. Gil, C.H. Lee et al.

- -

Computer Methods in Applied Mechanics and Engineering 392 (2022) 114680

$$\frac{\partial f}{\partial t} = \operatorname{div}_{\chi} \left( \boldsymbol{v} \otimes \boldsymbol{i} \right), \tag{18}$$

$$\frac{\partial \boldsymbol{h}}{\partial t} = \operatorname{curl}_{\boldsymbol{\chi}} \left( \boldsymbol{v} \times \boldsymbol{f} \right), \tag{19}$$

$$\frac{\partial j}{\partial t} = \operatorname{div}_{\chi} \left( \boldsymbol{h}^T \boldsymbol{v} \right).$$
<sup>(20)</sup>

Here,  $\operatorname{div}_{\chi}$  and  $\operatorname{curl}_{\chi}$  represent the divergence and curl operators carried out with respect to the referential configuration, and the referential identity tensor *i* is being defined as  $i = \sum_{i=1}^{3} e_{\chi}^{i} \otimes e_{\chi}^{i}$  with

$$\boldsymbol{e}_{\boldsymbol{\chi}}^{1} = \begin{bmatrix} 1\\0\\0 \end{bmatrix}; \quad \boldsymbol{e}_{\boldsymbol{\chi}}^{2} = \begin{bmatrix} 0\\1\\0 \end{bmatrix}; \quad \boldsymbol{e}_{\boldsymbol{\chi}}^{3} = \begin{bmatrix} 0\\0\\1 \end{bmatrix}.$$
(21)

It is worth emphasising that the primary unknown variables of the Updated Reference Lagrangian system introduced herein are  $\{p_{\chi}, f, h, j\}$ . Crucially, the updates of incremental geometric deformations  $\{f, h\}$  (18) and (19) must ensure the satisfaction of appropriate involutions, namely

$$\operatorname{curl}_{\mathbf{x}} f = \mathbf{0}; \qquad \operatorname{div}_{\mathbf{x}} h = \mathbf{0}. \tag{22}$$

Combining above equations into a system of conservation laws formulated in the reference configuration gives

$$\frac{\partial \mathcal{U}_{\chi}}{\partial t} + \sum_{i=1}^{3} \frac{\partial \mathcal{F}_{\chi}^{i}}{\partial \chi_{i}} = \mathcal{S}_{\chi}; \quad \text{in} \quad \Omega_{\chi}.$$
(23)

Here,  $\mathcal{U}_{\chi}$  is the vector of conservation variables (per unit of reference configuration),  $\mathcal{F}_{\chi}^{i}$  is the flux vector in *i*th direction at reference domain and  $\mathcal{S}_{\chi}$  is the source term (per unit of reference configuration). Their respective components are

$$\mathcal{U}_{\chi} = \begin{bmatrix} \boldsymbol{p}_{\chi} \\ \boldsymbol{f} \\ \boldsymbol{h} \\ \boldsymbol{j} \end{bmatrix}, \qquad \mathcal{F}_{\chi}^{i} = -\begin{bmatrix} \boldsymbol{\sigma}_{\chi} \boldsymbol{e}_{\chi}^{i} \\ \boldsymbol{v} \otimes \boldsymbol{e}_{\chi}^{i} \\ \boldsymbol{f} \times (\boldsymbol{v} \otimes \boldsymbol{e}_{\chi}^{i}) \\ \boldsymbol{h} : (\boldsymbol{v} \otimes \boldsymbol{e}_{\chi}^{i}) \end{bmatrix}, \qquad \mathcal{S}_{\chi} = \begin{bmatrix} \boldsymbol{f}_{\chi} \\ \boldsymbol{0} \\ \boldsymbol{0} \\ \boldsymbol{0} \end{bmatrix}.$$
(24)

It is important to remark the simplicity of the above system of Updated Reference Lagrangian conservation equations ((24)) with striking similarities to the previously presented Total Lagrangian system ((2)), which will naturally facilitate its computational implementation.

#### 2.3. Hamiltonian (or total energy density) and constitutive relationship for stresses

In order to provide a proper physical meaning to the conjugate fields (with respect to the conservation variables  $\mathcal{U}_{\chi}$ ) of the system at hand, consider the total energy density (or Hamiltonian  $\mathcal{H}_{\chi}$  [61–63] per unit of reference volume in case of reversible processes) defined by

$$\mathcal{H}_{\chi}(\chi, t) = \hat{\mathcal{H}}_{\chi}(\boldsymbol{p}_{\chi}, \boldsymbol{f}, \boldsymbol{h}, j) = \frac{1}{2\rho_{\chi}}\boldsymbol{p}_{\chi} \cdot \boldsymbol{p}_{\chi} + W_{\chi}(\boldsymbol{f}, \boldsymbol{h}, j),$$
(25)

and  $\mathcal{H}_{\chi}(\chi, t)$  and  $\hat{\mathcal{H}}_{\chi}(p_{\chi}, f, h, j)$  represent alternative functional representations of the same magnitude. For the isothermal case under consideration, the notion of Hamiltonian is simply the total energy per unit reference volume. Making use of Eq. (25), and introducing the conjugate stresses with respect to  $\{f, h, j\}$  to be

$$\Sigma_f = \frac{\partial W_{\chi}}{\partial f}; \qquad \Sigma_h = \frac{\partial W_{\chi}}{\partial h}; \qquad \Sigma_j = \frac{\partial W_{\chi}}{\partial j},$$
(26)

the associated work conjugates  $\mathcal{V}_{\chi}$  can then be obtained as

$$\boldsymbol{\mathcal{V}}_{\boldsymbol{\chi}} = \frac{\partial \hat{\mathcal{H}}_{\boldsymbol{\chi}}}{\partial \boldsymbol{\mathcal{U}}_{\boldsymbol{\chi}}} = \begin{bmatrix} \frac{\partial \hat{\mathcal{H}}_{\boldsymbol{\chi}}}{\partial p_{\boldsymbol{\chi}}} \\ \frac{\partial \hat{\mathcal{H}}_{\boldsymbol{\chi}}}{\partial f} \\ \frac{\partial \hat{\mathcal{H}}_{\boldsymbol{\chi}}}{\partial h} \\ \frac{\partial \hat{\mathcal{H}}_{\boldsymbol{\chi}}}{\partial j} \end{bmatrix} = \begin{bmatrix} \boldsymbol{v} \\ \frac{\partial W_{\boldsymbol{\chi}}}{\partial f} \\ \frac{\partial W_{\boldsymbol{\chi}}}{\partial h} \\ \frac{\partial W_{\boldsymbol{\chi}}}{\partial j} \end{bmatrix} = \begin{bmatrix} \boldsymbol{v} \\ \boldsymbol{\Sigma}_{f} \\ \boldsymbol{\Sigma}_{h} \\ \boldsymbol{\Sigma}_{j} \end{bmatrix}.$$
(27)

Upon the use of the tensor cross product property shown in [50], the above conjugate stresses  $\{\Sigma_f, \Sigma_h, \Sigma_j\}$  can indeed be related to the standard Cauchy stress tensor  $\sigma_{\chi}$  via the rate of internal work per unit reference volume, described as

$$\boldsymbol{\sigma}_{\boldsymbol{\chi}} : \dot{\boldsymbol{f}} = \dot{W}_{\boldsymbol{\chi}}(\boldsymbol{f}, \boldsymbol{h}, j) \tag{28a}$$

$$= \Sigma_f : \dot{f} + \Sigma_h : \dot{h} + \Sigma_j \dot{j}$$
(28b)

$$= \Sigma_f : \dot{f} + \Sigma_h : \left( f \times \dot{f} \right) + \Sigma_j (h : \dot{f})$$
(28c)

$$= \left( \Sigma_f + \Sigma_h \times f + \Sigma_j h \right) : \dot{f}, \tag{28d}$$

where the superimposed dot represents the usual material time derivative describing the change in f associated with a specific particle initially located at position X. Comparing the left hand side of Eq. (28a) with Eq. (28d) leads to the following relationship

$$\sigma_{\chi} = \Sigma_f + \Sigma_h \times f + \Sigma_j h. \tag{29}$$

**Remark 2.** To complete the definition of the Cauchy stress tensor (29), one plausible option is to rewrite the referential conjugate stresses { $\Sigma_f$ ,  $\Sigma_h$ ,  $\Sigma_j$ } of the system in terms of their material counterparts { $\Sigma_F$ ,  $\Sigma_H$ ,  $\Sigma_J$ } (previously presented in Remark 1). This is achieved by first utilising the pull back equivalent of the convex multivariable function  $W_{\chi}(f, h, j) = J_{\chi}^{-1}W(F, H, J)$ , followed by the use of the multiplicative decomposition for the triplet of deformation measures (9) and the chain rule. For instance, the conjugate stress of f gives

$$\Sigma_{f} = \frac{\partial W_{\chi}(f, h, j)}{\partial f} = J_{\chi}^{-1} \frac{\partial W\left(F(fF_{\chi}), H(hH_{\chi}), J(jJ_{\chi})\right)}{\partial f}$$
(30a)

$$= J_{\chi}^{-1} \frac{\partial W}{\partial F} : \frac{\partial F(fF_{\chi})}{\partial f}$$
(30b)

$$= J_{\chi}^{-1} \boldsymbol{\Sigma}_{\boldsymbol{F}} \boldsymbol{F}_{\chi}^{T}.$$
(30c)

Similarly, the conjugate stresses of h and j are now given by

$$\boldsymbol{\Sigma}_{\boldsymbol{h}} = J_{\boldsymbol{\chi}}^{-1} \boldsymbol{\Sigma}_{\boldsymbol{H}} \boldsymbol{H}_{\boldsymbol{\chi}}^{T}; \qquad \boldsymbol{\Sigma}_{j} = \boldsymbol{\Sigma}_{J}.$$
(31)

**Remark 3.** Many practical engineering applications often exhibit some permanent inelastic deformation. In order to describe this irrecoverable behaviour, the simple case of rate-independent von Mises plasticity in conjunction with either linear or nonlinear hardening rule will be considered in this paper. Within the context of large strains, it is customary to decompose the deformation gradient tensor F into an elastic component  $F_e$  and a permanent deformation component  $F_p$ , namely  $F = F_e F_p$ . In addition, and utilising the decomposition of  $F = f F_{\chi}$ , the elastic left Cauchy Green tensor  $b_e$  can now be rewritten in terms of both incremental deformation gradient tensor f and referential plastic strain  $c_p$  as

$$\boldsymbol{b}_{e} = \boldsymbol{F}\boldsymbol{C}_{p}^{-1}\boldsymbol{F}^{T} = \boldsymbol{f}\left(\boldsymbol{F}_{\chi}\boldsymbol{C}_{p}^{-1}\boldsymbol{F}_{\chi}^{T}\right)\boldsymbol{f}^{T} = \boldsymbol{f}\boldsymbol{c}_{p}^{-1}\boldsymbol{f}^{T}; \qquad \boldsymbol{c}_{p}^{-1} = \boldsymbol{F}_{\chi}\boldsymbol{C}_{p}^{-1}\boldsymbol{F}_{\chi}^{T}.$$
(32)

The standard return mapping algorithm to update the (referential) plastic strain  $c_p$  in order to ensure that the Kirchhoff stress satisfies, for instance, a von Mises type of plastic constraint can be seen in Ref. [26].

## 3. Weak form statements

In general, a standard weak variational statement for the isothermal mechanical system is established by multiplying the local differential Eqs. (23) (written in terms of the conservation variables  $\mathcal{U}_{\chi}$ ) with their appropriate work conjugate virtual fields  $\delta \mathcal{V}_{\chi}$ , and integrating over the reference domain  $\Omega_{\chi}$  of the body, to give

$$0 = \int_{\Omega_{\chi}} \delta \boldsymbol{\mathcal{V}}_{\chi} \bullet \frac{\partial \boldsymbol{\mathcal{U}}_{\chi}}{\partial t} d\Omega_{\chi} - \int_{\Omega_{\chi}} \delta \boldsymbol{\mathcal{V}}_{\chi} \bullet \boldsymbol{\mathcal{S}}_{\chi} d\Omega_{\chi} + \int_{\Omega_{\chi}} \delta \boldsymbol{\mathcal{V}}_{\chi} \bullet \frac{\partial \boldsymbol{\mathcal{F}}_{i}^{\chi}}{\partial \chi_{i}} d\Omega_{\chi},$$
(33)

where the symbol  $\bullet$  is used to denote the inner (dual) product of work conjugate pairs. Integrating by parts the last term on the right hand side of (33), and re-arranging the resulting equation yields

$$\int_{\Omega_{\chi}} \delta \mathcal{V}_{\chi} \bullet \frac{\partial \mathcal{U}_{\chi}}{\partial t} d\Omega_{\chi} = \int_{\Omega_{\chi}} \delta \mathcal{V}_{\chi} \bullet \mathcal{S}_{\chi} d\Omega_{\chi} + \int_{\Omega_{\chi}} \mathcal{F}_{i}^{\chi} \bullet \frac{\partial \delta \mathcal{V}_{\chi}}{\partial \chi_{i}} d\Omega_{\chi} - \int_{\partial \Omega_{\chi}} \delta \mathcal{V}_{\chi} \bullet \mathcal{F}_{N}^{\chi} dA^{\chi},$$
(34)

where the normal fluxes are defined as  $\mathcal{F}_N^{\chi} = \mathcal{F}_i^{\chi} N_i^{\chi}$  with  $N_i^{\chi}$  being the outward unit normal to the reference domain in the *i*th direction. Above representation (34) can be particularised to the case of the linear momentum  $p_{\chi}$  and the triplet of incremental geometric deformation measures  $\{f, h, j\}$  as

$$\int_{\Omega_{\chi}} \delta \boldsymbol{v} \cdot \frac{\partial \boldsymbol{p}_{\chi}}{\partial t} d\Omega_{\chi} = -\int_{\Omega_{\chi}} \boldsymbol{\sigma}_{\chi} : \boldsymbol{\nabla}_{\chi} \delta \boldsymbol{v} d\Omega_{\chi} + \int_{\Omega_{\chi}} \delta \boldsymbol{v} \cdot \boldsymbol{f}_{\chi} d\Omega_{\chi} + \int_{\partial \Omega_{\chi}} \delta \boldsymbol{v} \cdot \boldsymbol{t}_{B} dA^{\chi};$$
(35a)

$$\int_{\Omega_{\chi}} \delta \boldsymbol{\Sigma}_{f} : \frac{\partial f}{\partial t} d\Omega_{\chi} = \int_{\Omega_{\chi}} \delta \boldsymbol{\Sigma}_{f} : \nabla_{\chi} \boldsymbol{v} d\Omega_{\chi};$$
(35b)

$$\int_{\Omega_{\chi}} \delta \boldsymbol{\Sigma}_{\boldsymbol{h}} : \frac{\partial \boldsymbol{h}}{\partial t} \, d\Omega_{\chi} = \int_{\Omega_{\chi}} \delta \boldsymbol{\Sigma}_{\boldsymbol{h}} : \left( \boldsymbol{f} \times \nabla_{\chi} \boldsymbol{v} \right) \, d\Omega_{\chi}; \tag{35c}$$

$$\int_{\Omega_{\chi}} \delta \Sigma_j \frac{\partial j}{\partial t} d\Omega_{\chi} = \int_{\Omega_{\chi}} \delta \Sigma_j \boldsymbol{h} : \nabla_{\chi} \boldsymbol{v} d\Omega_{\chi}.$$
(35d)

The main purpose of integrating by parts as shown above is to enable the imposition of the boundary conditions via boundary fluxes. This is indeed useful for the momentum update (35a) as it introduces naturally the boundary tractions  $t_B$ , but less so in the case of geometric conservation Eqs. (35b)–(35d).

#### 4. Spatial discretisation

## 4.1. A point of departure: corrected SPH approximation at material domain

Consider the elastic body discretised by a cloud of particles as shown in Fig. 2. In the context of corrected SPH methods [20,28], any arbitrary vector function g is in general interpolated at any given position (quadrature point) via corrected (scatter) kernel functions  $\tilde{W}$  with a given compact support of radius 2h around every particle. For a given position  $X_a$ , the vector g can be approximated as

$$g(X_{a}) = \sum_{b \in \Lambda_{a}^{b}} V_{b} g_{b} \tilde{W}_{b}(\|\boldsymbol{r}_{ab}\|, h_{b}); \qquad \tilde{W}_{b}(\|\boldsymbol{r}_{ab}\|, h_{b}) = \mathcal{C}_{*} \left[ W_{b}(\|\boldsymbol{r}_{ab}\|, h_{b}) \right],$$
(36)

where  $||\mathbf{r}_{ab}|| = \sqrt{\mathbf{r}_{ab} \cdot \mathbf{r}_{ab}}$  being the distance to its centre and  $\mathbf{r}_{ab} = \mathbf{X}_a - \mathbf{X}_b$  the position vector. Typically, the material smoothing length  $h_b$  is in the form of

$$h_{b}(\|\boldsymbol{r}_{bj}\|) = f_{h} \max_{j \in \Lambda_{b}^{j}} \left( \|\boldsymbol{r}_{bj}\| \right); \qquad \|\boldsymbol{r}_{bj}\| = \sqrt{\boldsymbol{r}_{bj} \cdot \boldsymbol{r}_{bj}}; \qquad \boldsymbol{r}_{bj} = \boldsymbol{X}_{b} - \boldsymbol{X}_{j},$$
(37)

where  $f_h$  is the (scalar) coefficient used to scale the size of the compact support. Here,  $\Lambda_a^b$  (or  $\Lambda_b^j$ ) represents the set of neighbouring particles b (or j) belonging to the domain of influence (or compact support) of a given radius 2hof particle a (or b),  $V_b$  and  $g_b$  represent the material volume and time-varying vector function g stored at particle b. In this paper, we employ the commonly used quadratic kernel function described in Ref. [64], with the value of scaling coefficient  $f_h = 0.6$ .

To ensure the exact satisfaction of both constant and linear completeness, above corrected kernel approximation (scalar) operator  $C_*[\bullet]$  is defined according to [28] as

$$\mathcal{C}_*\left[W_b(\|\boldsymbol{r}_{ab}\|, h_b)\right] = \alpha_a \left[1 + \boldsymbol{\beta}_a \cdot \boldsymbol{r}_{ab}\right] W_b(\|\boldsymbol{r}_{ab}\|, h_b), \tag{38}$$



Fig. 2. Illustration of the domain of influence (also known as compact support) for particle a.

with the parameters

$$\alpha_a = \left(\sum_{b \in \Lambda_a^b} V_b \left[1 + \boldsymbol{\beta}_a \cdot \boldsymbol{r}_{ab}\right] W_b(\|\boldsymbol{r}_{ab}\|, h_b)\right)^{-1}$$
(39)

and

$$\boldsymbol{\beta}_{a} = -\left[\sum_{b \in \Lambda_{a}^{b}} \left(V_{b}\boldsymbol{r}_{ab} \otimes \boldsymbol{r}_{ab}\right) W_{b}(\boldsymbol{X}_{a})\right]^{-1} \sum_{b \in \Lambda_{a}^{b}} V_{b}\boldsymbol{r}_{ab} W_{b}(\|\boldsymbol{r}_{ab}\|, h_{b}).$$
(40)

In addition, for the evaluation of the material gradient of any arbitrary vector function g, we employ the following approximation introduced in [28]

$$\boldsymbol{\nabla}_{0}\boldsymbol{g}(\boldsymbol{X}_{a}) = \sum_{b \in \Lambda_{a}^{b}} V_{b}\left(\boldsymbol{g}_{b} - \boldsymbol{g}_{a}\right) \otimes \tilde{\boldsymbol{\nabla}}_{0} W_{b}(\|\boldsymbol{r}_{ab}\|, h_{b}); \qquad \tilde{\boldsymbol{\nabla}}_{0} W_{b}(\|\boldsymbol{r}_{ab}\|, h_{b}) = \boldsymbol{\mathcal{C}}_{*}\left[\boldsymbol{\nabla}_{0} W_{b}(\|\boldsymbol{r}_{ab}\|, h_{b})\right].$$
(41)

Specifically, the gradient correction (vector) operator  $\mathcal{C}_*[\bullet]$  is defined as

$$\mathcal{C}_*\left[\nabla_0 W_b(\|\boldsymbol{r}_{ab}\|, h_b)\right] = -\left(\sum_{b \in \Lambda_a^b} V_b \nabla_0 W_b(\|\boldsymbol{r}_{ab}\|, h_b) \otimes \boldsymbol{r}_{ab}\right)^{-1} \nabla_0 W_b(\|\boldsymbol{r}_{ab}\|, h_b).$$
(42)

In expression (41), the term  $-g_a$  is included in order to ensure that the gradient vanishes for a uniform field [28]. In addition, the use of the kernel gradient correction  $\tilde{\nabla}_0$  ensures that the gradient of any linear field is exactly evaluated [10].

Given the fact that the kernel is usually a function of the distance ||r|| to its centre, the gradient evaluation can now follow via the chain rule, namely

$$\nabla_0 W_b(\|\boldsymbol{r}_{ab}\|, h_b) = \frac{dW(\|\boldsymbol{r}_{ab}\|, h_b)}{d\|\boldsymbol{r}_{ab}\|} \frac{d\|\boldsymbol{r}_{ab}\|}{d\boldsymbol{r}_{ab}} = \frac{dW(\|\boldsymbol{r}_{ab}\|, h_b)}{d\|\boldsymbol{r}_{ab}\|} \frac{\boldsymbol{r}_{ab}}{\|\boldsymbol{r}_{ab}\|}.$$
(43)

 $\tilde{\nabla}$ 

## 4.2. Corrected SPH approximation at reference domain

For the case of the Updated Reference Lagrangian formalism, the kernel function as well as its gradient evaluation must be expressed in terms of the reference configuration. Since we aim at applications involving large deformations, standard updated isotropic (spherical) kernels would not be optimal (see e.g. [65]). Due to the possible presence of strong anisotropic changes in volume, and consequently in the particle spacing, the use of anisotropic kernels is better suited. Moreover, in such approach, provided that no severe topological changes take place, the initial list of neighbouring particles can be kept constant throughout the simulation, providing exceptional accuracy in a cost-efficient manner. This can be achieved by relating quantities in the material and reference configurations via the concepts of *push forward*  $\phi_*[\bullet]$  and *pull back*  $\phi_*^{-1}[\bullet]$  operations. For example, the reference kernel function  $W_b(\chi_a)$  can be considered as the push-forward equivalent of the material kernel function. This can be expressed in terms of the operation

$$W_b(\boldsymbol{\chi}_a) = \phi_* \left[ W_b(\|\boldsymbol{r}_{ab}\|, h_b) \right] = J_{\boldsymbol{\chi}, b}^{-1} W_b(\|\boldsymbol{r}_{ab}\|, h_b).$$
(44)

In the same spirit of Eq. (44), the push-forward equivalent of the material kernel gradient is<sup>2</sup>

$$\nabla_{\boldsymbol{\chi}} W_b(\boldsymbol{\chi}_a) = \boldsymbol{\phi}_* \left[ \nabla_0 W_b(\|\boldsymbol{r}_{ab}\|, h_b) \right] = J_{\boldsymbol{\chi}, b}^{-1} \boldsymbol{F}_{\boldsymbol{\chi}, a}^{-T} \nabla_0 W_b(\|\boldsymbol{r}_{ab}\|, h_b).$$
(45)

Notice that the evaluations for the reference kernel function (44) and the associated gradient (45) still require geometrical information attached to the material domain, that is,  $r_{ab}$ . In order to ensure that the kernel function depends solely on the information at the reference domain, one viable option is to approximate the material vector used in (44) as the pull back equivalent of its reference counterpart, that is

$$\boldsymbol{r}_{ab} = \boldsymbol{X}_{a} - \boldsymbol{X}_{b} \approx \boldsymbol{\phi}_{*}^{-1} \left[ \underbrace{\boldsymbol{\chi}_{a} - \boldsymbol{\chi}_{b}}_{\boldsymbol{r}_{\boldsymbol{\chi},ab}} \right]$$

$$= \left[ \boldsymbol{F}_{\boldsymbol{\chi},ab}^{\text{Ave}} \right]^{-1} \boldsymbol{r}_{\boldsymbol{\chi},ab} = \hat{\boldsymbol{r}}_{ab}.$$
(46a)
(46b)

where  $\mathbf{r}_{\chi,ab} = \mathbf{\chi}_a - \mathbf{\chi}_b$  denotes the reference position vector whereas the average (second order) approximation is  $\mathbf{F}_{\chi,ab}^{\text{Ave}} = \frac{1}{2} \left[ \mathbf{F}_{\chi,a} + \mathbf{F}_{\chi,b} \right]$ . With this at hand, the reference kernel function can now be rewritten as

$$W(\|\boldsymbol{r}_{ab}\|, h_b) \approx W(\|\boldsymbol{\phi}_*^{-1}[\boldsymbol{r}_{\boldsymbol{\chi}, ab}]\|, h_b(\|\boldsymbol{\phi}_*^{-1}[\boldsymbol{r}_{\boldsymbol{\chi}, ab}]\|)) = W(\|\hat{\boldsymbol{r}}_{ab}\|, \hat{h}_b); \qquad \hat{h}_b = f_h \max_{j \in \Lambda_b^j} \left(\|\hat{\boldsymbol{r}}_{ab}\|\right).$$
(47)

Making use of (43) and the pull back quantities (46), we can further approximate the material gradient described in (45) to be

$$\nabla_0 W_b(\|\boldsymbol{r}_{ab}\|, h_b) \approx \frac{dW(\|\hat{\boldsymbol{r}}_{ab}\|, \hat{h}_b)}{d\|\hat{\boldsymbol{r}}_{ab}\|} \frac{\hat{\boldsymbol{r}}_{ab}}{\|\hat{\boldsymbol{r}}_{ab}\|} = \nabla_0 W_b(\|\hat{\boldsymbol{r}}_{ab}\|, \hat{h}_b).$$
(48)

In the current work, we exploit four different options to obtain reference kernel functions and their corresponding gradient evaluations, which are summarised as follows:

• **Option#1**. Both the kernel function and its gradient are first corrected (ensuring zeroth- and first-order completeness) in the material domain and then are pushed forward to the reference domain. Mathematically, these are expressed as

$$\tilde{W}_{b}(\boldsymbol{\chi}_{a}) = \phi_{*} \left[ \mathcal{C}_{*} \left[ W_{b}(\|\boldsymbol{r}_{ab}\|, h_{b}) \right] \right] = J_{\boldsymbol{\chi}, b}^{-1} \mathcal{C}_{*} \left[ W_{b}(\|\boldsymbol{r}_{ab}\|, h_{b}) \right];$$
(49a)

$${}_{\chi}W_{b}(\chi_{a}) = \phi_{*} \left[ \mathcal{C}_{*} \left[ \nabla_{0}W_{b}(\|\boldsymbol{r}_{ab}\|, h_{b}) \right] \right] = J_{\chi,b}^{-1} \boldsymbol{F}_{\chi,a}^{-T} \mathcal{C}_{*} \left[ \nabla_{0}W_{b}(\|\boldsymbol{r}_{ab}\|, h_{b}) \right].$$
(49b)

• **Option#2**. On the contrary, this particular option first pushes forward the material kernel function and its gradient to the reference domain, followed by the application of appropriate corrections at reference domain,

 $<sup>^2</sup>$  This is not a classical push forward transformation of a vector typically used in continuum mechanics. Notice that in this case, we define the push forward mappings of kernel approximation (44) and its gradient evaluation (45) between two different configurations, hence Jacobian is necessary to be included in those expressions.

that is

$$\tilde{W}_{b}(\boldsymbol{\chi}_{a}) = \mathcal{C}_{*} \left[ \phi_{*} \left[ W_{b}(\|\boldsymbol{r}_{ab}\|, h_{b}) \right] \right] = \mathcal{C}_{*} \left[ J_{\boldsymbol{\chi}, b}^{-1} W_{b}(\|\boldsymbol{r}_{ab}\|, h_{b}) \right];$$
(50a)

$$\tilde{\nabla}_{\boldsymbol{\chi}} W_b(\boldsymbol{\chi}_a) = \mathcal{C}_* \left[ \boldsymbol{\phi}_* \left[ \nabla_0 W_b(\|\boldsymbol{r}_{ab}\|, h_b) \right] \right] = \mathcal{C}_* \left[ J_{\boldsymbol{\chi}, b}^{-1} \boldsymbol{F}_{\boldsymbol{\chi}, a}^{-T} \nabla_0 W_b(\|\boldsymbol{r}_{ab}\|, h_b) \right].$$
(50b)

• **Option#3**. As an alternative to option#2, we can rewrite the material kernel function (47) (in terms of the reference geometrical information) via the pull back equivalent of the reference vector (46) to yield

$$\tilde{W}_b(\boldsymbol{\chi}_a) = \mathcal{C}_* \left[ \boldsymbol{\phi}_* \left[ W_b(\| \boldsymbol{\phi}_*^{-1} [\boldsymbol{r}_{\boldsymbol{\chi}, ab}] \|, h_b(\| \boldsymbol{\phi}_*^{-1} [\boldsymbol{r}_{\boldsymbol{\chi}, ab}] \|)) \right] \right]$$
(51a)

$$= C_* \left[ J_{\chi,b}^{-1} W_b(\|\hat{\boldsymbol{r}}_{ab}\|, \hat{h}_b) \right];$$
(51b)

$$\tilde{\boldsymbol{\nabla}}_{\boldsymbol{\chi}} W_b(\boldsymbol{\chi}_a) = \boldsymbol{\mathcal{C}}_* \left[ \boldsymbol{\phi}_* \left[ \boldsymbol{\nabla}_0 W_b(\| \boldsymbol{\phi}_*^{-1} [\boldsymbol{r}_{\boldsymbol{\chi}, ab}] \|, h_b \left( \| \boldsymbol{\phi}_*^{-1} [\boldsymbol{r}_{\boldsymbol{\chi}, ab}] \| \right) \right) \right]$$
(51c)

$$= \mathcal{C}_* \left[ J_{\boldsymbol{\chi},b}^{-1} F_{\boldsymbol{\chi},a}^{-T} \nabla_0 W_b(\|\hat{\boldsymbol{r}}_{ab}\|, \hat{h}_b) \right].$$
(51d)

• **Option#4.** Alternatively, we can re-define the push-forward operation of the gradient operator (45) using the average of the deformation mapping between pairwise interacting particles, that is  $\boldsymbol{\phi}_* [\nabla_0 W_b(\|\boldsymbol{r}_{ab}\|, h_b)] = J_{\boldsymbol{\chi}, b}^{-1} [\boldsymbol{F}_{\boldsymbol{\chi}, ab}^{\text{Ave}}]^{-T} \nabla_0 W_b(\|\boldsymbol{r}_{ab}\|, h_b)$ . This yields

$$\tilde{W}_b(\boldsymbol{\chi}_a) = \mathcal{C}_* \left[ \boldsymbol{\phi}_* \left[ W_b(\|\boldsymbol{\phi}_*^{-1}[\boldsymbol{r}_{\boldsymbol{\chi},ab}]\|, h_b(\|\boldsymbol{\phi}_*^{-1}[\boldsymbol{r}_{\boldsymbol{\chi},ab}]\|)) \right] \right]$$
(52a)

$$= \mathcal{C}_* \left[ J_{\boldsymbol{\chi},b}^{-1} W_b(\|\hat{\boldsymbol{r}}_{ab}\|, \hat{h}_b) \right];$$
(52b)

$$\tilde{\nabla}_{\boldsymbol{\chi}} W_b(\boldsymbol{\chi}_a) = \mathcal{C}_* \left[ \boldsymbol{\phi}_* \left[ \nabla_0 W_b(\|\boldsymbol{\phi}_a^{-1}[\boldsymbol{r}_{\boldsymbol{\chi},ab}]\|, h_b\left(\|\boldsymbol{\phi}_a^{-1}[\boldsymbol{r}_{\boldsymbol{\chi},ab}]\|\right)) \right] \right]$$
(52c)

$$= \mathcal{C}_* \left[ J_{\boldsymbol{\chi}, ab}^{-1} \left[ F_{\boldsymbol{\chi}, ab}^{\text{Ave}} \right]^{-T} \nabla_0 W_b(\|\hat{\boldsymbol{r}}_{ab}\|, \hat{h}_b) \right].$$
(52d)

For the numerical examples examined in this paper, the robustness and accuracy of the overall algorithm is not adversely affected regardless of the option used for SPH kernel approximation. However, we believe it is extremely informative to explore these various possibilities and provide the user with several options, in order to choose the most convenient one from the point of view of their computational implementation.

## 4.3. SPH semi-discrete equations

Upon the use of corrected SPH kernel approximations for  $\mathcal{U}_{\chi}$  and  $\delta \mathcal{V}_{\chi}$ , along with the corrected gradient evaluation for  $\nabla_{\chi} \delta \mathcal{V}_{\chi}$ , in expression (34), the SPH discretisation for the system { $p_{\chi}$ , f, h, j} described in (17)–(20) becomes

$$V_a^{\chi} \frac{d\boldsymbol{p}_{\chi}^a}{dt} = \boldsymbol{T}_a^{\chi} + V_a^{\chi} \boldsymbol{f}_{\chi}^a + A_a^{\chi} \boldsymbol{t}_B^a,$$
(53)

$$V_a^{\chi} \frac{df_a}{dt} = \sum_{b \in \Lambda_a^b} \frac{1}{2} (\boldsymbol{v}_b - \boldsymbol{v}_a) \otimes \boldsymbol{C}_{ab}^{\chi}, \tag{54}$$

$$V_{a}^{\chi} \frac{d\boldsymbol{h}_{a}}{dt} = \boldsymbol{f}_{a} \times \left( \sum_{b \in A_{a}^{b}} \frac{1}{2} (\boldsymbol{v}_{b} - \boldsymbol{v}_{a}) \otimes \boldsymbol{C}_{ab}^{\chi} \right),$$
(55)

$$V_a^{\mathbf{\chi}} \frac{dj_a}{dt} = \mathbf{h}_a : \left( \sum_{b \in \Lambda_a^b} \frac{1}{2} (\mathbf{v}_b - \mathbf{v}_a) \otimes \mathbf{C}_{ab}^{\mathbf{\chi}} \right) + \sum_{b \in \Lambda_a^b} \mathcal{D}_{ab}^j.$$
(56)

Here, the representation for the (algorithmic) internal nodal force  $T_a^{\chi}$  is defined as

$$\boldsymbol{T}_{a}^{\boldsymbol{\chi}} = \sum_{b \in \Lambda_{a}^{b}} \frac{1}{2} \left( \boldsymbol{\sigma}_{\boldsymbol{\chi}}^{\boldsymbol{\chi}} \boldsymbol{C}_{ab}^{\boldsymbol{\chi}} - \boldsymbol{\sigma}_{\boldsymbol{\chi}}^{b} \boldsymbol{C}_{ba}^{\boldsymbol{\chi}} \right) + \sum_{b \in \Lambda_{a}^{b}} \boldsymbol{\mathcal{D}}_{ab}^{\boldsymbol{p}_{\boldsymbol{\chi}}}, \tag{57}$$

with the (pseudo) area operators, see [26], defined as

$$\boldsymbol{C}_{ab}^{\boldsymbol{\chi}} = 2V_{a}^{\boldsymbol{\chi}}V_{b}^{\boldsymbol{\chi}}\tilde{\boldsymbol{\nabla}}_{\boldsymbol{\chi}}W_{b}(\boldsymbol{\chi}_{a}), \qquad \boldsymbol{C}_{ba}^{\boldsymbol{\chi}} = 2V_{a}^{\boldsymbol{\chi}}V_{b}^{\boldsymbol{\chi}}\tilde{\boldsymbol{\nabla}}_{\boldsymbol{\chi}}W_{a}(\boldsymbol{\chi}_{b}).$$
(58)

In the above expressions,  $V_a^{\chi}$  and  $A_a^{\chi}$  represent the volume and the referential tributary area and  $t_B^a$  its traction vector computed directly from the given traction boundary conditions. Notice that  $A_a^{\chi} = 0$  for those particles not placed on the boundary.

Finally, the remaining terms  $\{\mathcal{D}_{ab}^{p_{\chi}}, \mathcal{D}_{ab}^{j}\}$  in Eqs. (53) and (56) are the so-called numerical dissipation terms. These terms can be introduced and derived using the semi-discrete version of the classical Coleman–Noll procedure in order to ensure the production of non-negative numerical entropy. Following a similar procedure presented in Ref. [66], and after some algebraic manipulations, the Godunov-type numerical dissipations are summarised here for completeness

$$\mathcal{D}_{ab}^{p_{\chi}} = S_{ab}^{p_{\chi}}(\boldsymbol{v}_{ab}^{R} - \boldsymbol{v}_{ab}^{L}); \qquad \mathcal{D}_{ab}^{j} = S_{ab}^{j}\left(\boldsymbol{\Sigma}_{j,ab}^{R} - \boldsymbol{\Sigma}_{j,ab}^{L}\right),$$
(59)

where  $[\bullet]_{ab}^{L,R}$  are the left and the right states of the reconstructed values obtained via linear reconstruction procedure [26]. The positive definite stabilisation tensors are expressed in terms of the average pressure wave speed  $c_{p,ab}^{Ave}$  and the average shear wave speed  $c_{s,ab}^{Ave}$  as

$$\boldsymbol{S}_{ab}^{\boldsymbol{p}_{\boldsymbol{\chi}}} = \frac{\rho_{\boldsymbol{\chi},ab}^{\text{Ave}} \|\boldsymbol{C}_{ab}^{\boldsymbol{\chi},\text{Skew}}\|}{2} \left[ c_{p,ab}^{\text{Ave}} \boldsymbol{n}_{ab} \otimes \boldsymbol{n}_{ab} + c_{s,ab}^{\text{Ave}} (\boldsymbol{I} - \boldsymbol{n}_{ab} \otimes \boldsymbol{n}_{ab}) \right]; \qquad \boldsymbol{S}_{ab}^{j} = \frac{\boldsymbol{c}_{ab}^{\text{Skew}} \cdot \boldsymbol{c}_{ab}^{\text{Skew}}}{2\rho_{\boldsymbol{\chi},ab}^{\text{Ave}} c_{p,ab}^{\text{Ave}} \|\boldsymbol{C}_{ab}^{\boldsymbol{\chi},\text{Skew}}\|}, \tag{60}$$

where  $[\bullet]_{ab}^{Ave} = \frac{1}{2} ([\bullet]_a + [\bullet]_b)$  and the unit vector is given by  $\mathbf{n}_{ab} = \frac{\mathbf{x}_b - \mathbf{x}_a}{\|\mathbf{x}_b - \mathbf{x}_a\|}$ . The pseudo-area vector  $C_{ab}^{\chi, \text{Skew}}$  (along with its norm magnitude  $\|C_{ab}^{\chi, \text{Skew}}\|$ ) and its push forward equivalent (spatial) vector  $\mathbf{c}_{ab}^{\text{Skew}}$  are defined as  $C_{ab}^{\chi, \text{Skew}} = \frac{1}{2} (C_{ab}^{\chi} - C_{ba}^{\chi})$  and  $\mathbf{c}_{ab}^{\text{Skew}} = \frac{1}{2} (\mathbf{h}_a C_{ab}^{\chi} - \mathbf{h}_b C_{ba}^{\chi})$ . Notice here that the stabilisation term applied to the linear momentum evolution (53) alleviates the appearance

Notice here that the stabilisation term applied to the linear momentum evolution (53) alleviates the appearance of spurious zero energy modes due to rank deficiency inherent to the use of nodal particle integration, whereas the stabilisation in the volume map update (56) may potentially be used to address pressure instabilities (especially in near incompressibility) if necessary. For all the numerical examples presented below, it is worth pointing out that no dissipation has been included in (56) by strongly enforcing the value of  $\mathcal{D}_{ab}^{j}$  equal to zero.

#### 5. Time discretisation

Insofar as the resulting set of semi-discrete equations is rather large, it will only be appropriate to employ an explicit type of time integrator. In this work, a three stage Runge–Kutta explicit time integrator [67] is used. This is described by the following time update equations from time step  $t^n$  to  $t^{n+1}$ 

$$\mathcal{U}_{\chi,a}^{\star} = \mathcal{U}_{\chi,a}^{n} + \Delta t \, \dot{\mathcal{U}}_{\chi,a}^{n} \left( \mathcal{U}_{\chi,a}^{n} \right), \tag{61a}$$

$$\mathcal{U}_{\boldsymbol{\chi},a}^{\star\star} = \frac{3}{4} \mathcal{U}_{\boldsymbol{\chi},a}^{n} + \frac{1}{4} \left( \mathcal{U}_{\boldsymbol{\chi},a}^{\star} + \Delta t \, \dot{\mathcal{U}}_{\boldsymbol{\chi},a}^{\star} \left( \mathcal{U}_{\boldsymbol{\chi},a}^{\star} \right) \right), \tag{61b}$$

$$\mathcal{U}_{\boldsymbol{\chi},a}^{n+1} = \frac{1}{3} \mathcal{U}_{\boldsymbol{\chi},a}^{n} + \frac{2}{3} \left( \mathcal{U}_{\boldsymbol{\chi},a}^{\star\star} + \Delta t \, \dot{\mathcal{U}}_{\boldsymbol{\chi},a}^{\star\star} \left( \mathcal{U}_{\boldsymbol{\chi},a}^{\star\star} \right) \right). \tag{61c}$$

Additionally, the geometry is also updated via the same time integrator presented above, resulting in a monolithic time update procedure in which the unknowns  $\mathcal{U}_{\chi} = (p_{\chi}, f, h, j)^T$  are updated along with the (spatial) geometry x through Eqs. (61). Moreover, the maximum time step  $\Delta t = t^{n+1} - t^n$  is governed by a standard Courant-Friedrichs-Lewy (CFL) [47] condition given by

$$\Delta t = \alpha_{CFL} \min\left(\frac{\|\boldsymbol{x}_a - \boldsymbol{x}_b\|}{c_{p,ab}^{\text{Ave}}}\right),\tag{62}$$

where  $\alpha_{CFL}$  is the CFL stability number and  $c_{p,ab}^{Ave}$  the average pressure wave speed between particles *a* and *b*. Unless otherwise stated, a value of  $\alpha_{CFL} = 0.9$  has been chosen in the following examples to ensure both accuracy and stability.

It is worth noticing that in expression (62), the characteristic particle spacing  $||\mathbf{x}_a - \mathbf{x}_b||$  is computed with respect to the current (or spatial) configuration. The evaluation for pressure wave speed is however approximated by dividing the standard linear wave speed  $c_p^{\text{Lin}}$  by the average (minimum) stretch between the pairwise interacting particles *a* 

P.R.R. de Campos, A.J. Gil, C.H. Lee et al.

and b, described as

$$c_{p,ab}^{\text{Ave}} = \frac{c_{p}^{\text{Lin}}}{\lambda_{ab}^{\text{Ave}}}; \qquad c_{p}^{\text{Lin}} = \sqrt{\frac{\lambda + 2\mu}{\rho_{R}}}; \qquad \lambda_{ab}^{\text{Ave}} = \frac{1}{2} \left(\lambda_{a} + \lambda_{b}\right), \tag{63}$$

where  $\mu$  is the shear modulus,  $\lambda$  is the Lamé first parameter and  $\lambda_a$  and  $\lambda_b$  are the minimum stretch of particles *a* and *b*, respectively.

#### 6. Conservation of angular momentum

The resulting SPH algorithm does not exactly fulfil conservation of angular momentum, since the strain measures  $\{f, h, j\}$  are no longer exclusively obtained from the current geometry. Although deviation from strict conservation is minimal, a monolithic discrete angular momentum projection algorithm is presented, following the work of [24,68]. In our numerical experiments, we have observed that activation of this algorithm is not strictly necessary unless "machine accurate" strict angular momentum conservation is sought. Specifically, the local internal nodal force  $T_a^{\chi}$  is suitably modified (in a least-square sense) in order to preserve the global angular momentum, whilst still ensuring the global conservation of linear momentum.

Adapting the procedure presented in [68] to the Runge Kutta time integrator considered herein, sufficient conditions for the global preservation of the discrete linear and angular momentum within a time step are explicitly enforced at each of the three stages described as

$$\sum_{a} V_{a}^{\chi} \boldsymbol{T}_{a}^{\chi,\alpha} = \boldsymbol{0}; \qquad \sum_{a} V_{a}^{\chi} \boldsymbol{\mathfrak{X}}_{a}^{\alpha} \times \boldsymbol{T}_{a}^{\chi,\alpha} = \boldsymbol{0}; \qquad \boldsymbol{\mathfrak{X}}_{a}^{\alpha} = \begin{cases} \boldsymbol{\mathfrak{X}}_{a}^{n} - \Delta t \, \boldsymbol{v}_{a}^{n}; & \alpha = n \\ \boldsymbol{\mathfrak{X}}_{a}^{n} + \frac{\Delta t}{6} \left( \boldsymbol{v}_{a}^{n} - \boldsymbol{v}_{a}^{n} \right); & \alpha = * \\ \boldsymbol{\mathfrak{X}}_{a}^{n} + \frac{\Delta t}{6} \left( \boldsymbol{v}_{a}^{n} + \boldsymbol{v}_{a}^{\star} + 4 \boldsymbol{v}_{a}^{\star \star} \right); & \alpha = ** \end{cases}$$
(64)

A least-square minimisation procedure is used to obtain a modified set of internal nodal forces  $\hat{T}_a^{\chi}$  that satisfy the above conditions. This can be achieved by computing the minimum of the following functional (ignoring time arguments for brevity)

$$\Pi(\hat{\boldsymbol{T}}_{a}^{\boldsymbol{\chi}},\boldsymbol{\lambda}_{\mathrm{ang}},\boldsymbol{\lambda}_{\mathrm{lin}}) = \frac{1}{2} \sum_{a} V_{a}^{\boldsymbol{\chi}}(\hat{\boldsymbol{T}}_{a}^{\boldsymbol{\chi}} - \boldsymbol{T}_{a}^{\boldsymbol{\chi}}) \cdot (\hat{\boldsymbol{T}}_{a}^{\boldsymbol{\chi}} - \boldsymbol{T}_{a}^{\boldsymbol{\chi}}) - \boldsymbol{\lambda}_{\mathrm{ang}} \cdot \left(\sum_{a} V_{a}^{\boldsymbol{\chi}} \boldsymbol{\mathfrak{X}}_{a} \times \hat{\boldsymbol{T}}_{a}^{\boldsymbol{\chi}}\right) - \boldsymbol{\lambda}_{\mathrm{lin}} \cdot \left(\sum_{a} V_{a}^{\boldsymbol{\chi}} \hat{\boldsymbol{T}}_{a}^{\boldsymbol{\chi}}\right).$$

$$(65)$$

After some simple algebra, a modified set of internal nodal forces  $\hat{T}_a$  is obtained in the form of

$$\hat{\boldsymbol{T}}_{a}^{\boldsymbol{\chi}} = \boldsymbol{T}_{a}^{\boldsymbol{\chi}} + \boldsymbol{\lambda}_{\text{ang}} \times \boldsymbol{\mathfrak{X}}_{a} + \boldsymbol{\lambda}_{\text{lin}}, \tag{66}$$

where the Lagrange multipliers  $\{\lambda_{ang}, \lambda_{lin}\}$  are the solution of the following system of equations

$$\begin{bmatrix} \sum_{a} V_{a}^{\chi} \left[ (\mathbf{X}_{a} \cdot \mathbf{X}_{a}) \mathbf{I} - \mathbf{X}_{a} \otimes \mathbf{X}_{a} \right] & \sum_{a} V_{a}^{\chi} \hat{\mathbf{X}}_{a} \\ \sum_{a} V_{a}^{\chi} \hat{\mathbf{X}}_{a} & -\sum_{a} V_{a}^{\chi} \end{bmatrix} \begin{bmatrix} \mathbf{\lambda}_{ang} \\ \mathbf{\lambda}_{lin} \end{bmatrix} = \begin{bmatrix} -\sum_{a} V_{a}^{\chi} \mathbf{X}_{a} \times \mathbf{T}_{a}^{\chi} \\ \sum_{a} V_{a}^{\chi} \mathbf{T}_{a}^{\chi} \end{bmatrix},$$
(67)

with the indicial notation  $\left[\hat{\mathbf{X}}_{a}\right]_{ik} = \mathcal{E}_{ijk} \left[\mathbf{X}_{a}\right]_{j}$ .

## 7. Algorithmic description

For ease of understanding, this section summarises the algorithmic implementation of the SPH methodology presented in previous sections. We start by including a graphical representation of the method, as illustrated in Fig. 3.

Algorithm (1) summarises the complete algorithmic description of the proposed Updated Reference Lagrangian SPH methodology. Notice that, under certain conditions, the proposed methodology degenerates to the case of either Total Lagrangian SPH [26] or pure Updated Lagrangian SPH. For example, if the conditional statement in Algorithm 1 is set to "FALSE" for every time integration process, the reference domain would remain unchanged throughout the entire simulation. In this case, the reference domain coincides with the material domain, implying that



**Fig. 3.** The proposed Updated Reference Lagrangian algorithm comprises a series of multiplicative incremental configurations as illustrated above. For instance, from time  $t_n$  to  $t_{2n}$ , the complete deformation gradient at time  $t_{2n}$ , namely  $F_{2n}$ , is evaluated via the previously computed deformation gradient  $F_n$  (now being stored and assigned as a deformation gradient at intermediate configuration  $\Omega_v(t_n)$ ) and a series of incremental deformation gradients  $\{f_{\Delta_1}, f_{\Delta_2}, \dots, f_{\Delta_n}\}$  between the reference configuration  $\Omega_v(t_n)$  and the new configuration  $\Omega_v(t_{2n})$ .

the corrected kernel and gradient approximations are only carried out once. On the other hand, if the conditional statement returns "TRUE" at every time step of the time integration process, pure Updated Lagrangian SPH is retrieved. Specifically, corrections for kernel function and its gradient are performed at every time step, which may significantly increase the overall computational cost. One key feature of the proposed approach is the ability to suitably update the reference domain when certain criteria (e.g. internal state variable, maximum value of stretch) are met. This will be explored in forthcoming publications, especially in the application of fast dynamic fracture. However, in this work, we choose to update the reference domain at a fixed number n of time steps (pre-defined by user) with the aim to check the performance of this approach. Its graphical representation is illustrated in Fig. 3 for clarity.

## 8. Numerical examples

#### 8.1. Column

As a first example, a unit squared cross section column of length L = 6 m is studied. To assess the performance of the proposed method on various deformation modes (e.g. stretching, bending and twisting), the column is initiated with three different dynamic loading scenarios (see Fig. 4). These include:

- Scenario I: tensile mode. The column is pulled with an initial (linear) velocity profile described as  $v_0 = \left[0, \frac{V_0Y}{L}, 0\right]^T$  with the value of  $V_0 = 50 \text{ ms}^{-1}$ . The bottom surface of the column is clamped, the top surface is left free and the remaining boundaries are enforced with symmetric roller support.
- Scenario II: bending mode. The column is subjected to bending by the application of an initial linearly varying velocity profile given by  $\boldsymbol{v}_0 = \left[\frac{V_0Y}{L}, 0, 0\right]^T$  where  $V_0 = 10 \text{ ms}^{-1}$  is the maximum velocity applied. The column is clamped at the bottom and free on all other sides.

# Algorithm 1: Updated Reference Lagrangian SPH Algorithm

**Input** : initial geometry  $X_a$  and initial states of  $p_{\chi,a}$ ,  $f_a$ ,  $h_a$   $j_a$ **Output:** current geometry  $x_a$ , particle velocity  $v_a$  and current states of  $F_a$ ,  $H_a$  and  $J_a$ (1) INITIALISE  $F_{\chi,a} = H_{\chi,a} = I$ ,  $J_{\chi,a} = 1$  and  $x_a = \chi_a = X_a$ (2) FIND neighbouring particles within a given support size  $(\Lambda_a^b)$ (3) COMPUTE corrected kernel and gradient approximations for Time  $t_0$  to Time t do if update at this step = TRUE then (4) COMPUTE the velocity as  $\boldsymbol{v}_a = \frac{J_{\boldsymbol{\chi},a}\boldsymbol{p}_{\boldsymbol{\chi},a}}{\rho_R}$ (5) ASSIGN  $F_{\chi,a} \leftarrow f_a F_{\chi,a}$ ,  $H_{\chi,a} \leftarrow h_a H_{\chi,a}$ ,  $J_{\chi,a} \leftarrow j_a J_{\chi,a}$  and  $\chi_a \leftarrow x_a$ (6) UPDATE the linear momentum  $p_{\chi,a} = J_{\chi,a}^{-1} \rho_R v_a$ (7) REINITIALISE  $f_a$ ,  $h_a$ ,  $j_a$ (8) COMPUTE corrected kernel and gradient approximations (9) COMPUTE  $\sigma_{\chi,a}$ end (10) EVALUATE p and s-wave speeds:  $c_p$ ,  $c_s$ (11) COMPUTE time increment:  $\Delta t$ for *RK* time integrator = 1 to 3 do (12) COMPUTE slope of linear reconstruction procedure (13) COMPUTE right-hand-side of the mixed-based system:  $\dot{\boldsymbol{p}}_{\boldsymbol{\chi},a}, \, \boldsymbol{f}_{a}, \, \boldsymbol{h}_{a} \text{ and } \boldsymbol{j}_{a}$ (14) ENSURE conservation of angular momentum (15) COMPUTE smoothed velocities using the corrected kernel (16) EVOLVE  $p_{x,a}$ ,  $f_a$ ,  $h_a$   $j_a$  and  $x_a$ (17) COMPUTE  $\sigma_{\chi,a}$ end (18) COMPUTE smoothed variables using the corrected kernel (19) EXPORT results for this time step (20) ADVANCE in time end

• Scenario III: twisting mode. The column is twisted with an initial sinusoidal velocity field relative to the origin given by  $\mathbf{v}_0 = [0, \Omega_0 \sin(\pi^{\gamma/2L}), 0]^T$  where  $\Omega_0 = 105$  rad s<sup>-1</sup> represents the magnitude of initial angular velocity. Enforcement of boundary conditions is identical to that of Scenario II (bending mode).

Aiming to display convergence of the algorithm, four successively refined (uniform) models are used. These include (M1) 5  $\times$  5  $\times$  25, (M2) 7  $\times$  7  $\times$  35, (M3) 9  $\times$  9  $\times$  49 and (M4) 17  $\times$  17  $\times$  97 number of particles. In terms of constitutive model, a neo-Hookean material is used. A summary of the data simulation used is also presented in Table 1.



Fig. 4. Column test cases. (a) Tensile. (b) Bending. (c) Twisting.

Column: material and da	ta simulation used.		
Material density	$\rho_R$	1100	kg m <sup>-3</sup>
Young's modulus	Ε	0.017	GPa
Poisson's ratio	ν	0.3	
Kernel type	Quadratic kernel function		
Smoothing factor	$f_h$	0.6	
CFL	$\alpha_{\rm CFL}$	0.9	

## 8.1.1. Scenario I: Tensile mode

Table 1

This problem is specifically designed to check if the proposed method can circumvent spurious modes in stretching dominated problems. In this case, four different options for gradient corrections (summarised in Section 4.2) ensuring zeroth and first order completeness are employed and compared. As shown in Fig. 5, all four options described above yield extremely similar results in comparison to the published Total Lagrangian result [26], almost undistinguishable. Moreover, the global numerical dissipation introduced in the algorithm can be assessed by computing the difference between the total conserved energy and the summation of kinetic and elastic strain energies, as reported for model M1 in Fig. 6. It is interesting to notice that all four possible options ensure the generation of global numerical dissipation over the entire duration of the simulation. Only slight variations are observed. Regardless of which option is used, the total numerical dissipation of the algorithm is reduced when increasing the number of particles. This is shown in Fig. 7. It is worth pointing out that, for problems accompanied with large topological changes (such as material separation in dynamic fracture), it is instructive to use **Option#3** and **Option#4** where the computations of kernel function and its gradient are carried out at (post-fractured) referential configuration. This is in contrast to **Option#1** and **Option#2** where the computations are at (pre-fractured) material configuration without considering separation process, hence not suitable for scenario involving severe topological changes.

#### 8.1.2. Scenario II : Bending mode

The main objective of this problem [24,69,70] is to assess the applicability of the proposed algorithm by carrying out frequent updates of the reference configuration. Doing this with the classical SPH algorithm [23] would activate spurious mechanisms, and eventually lead to the breakdown of the scheme. For this reason, updates of the reference configuration are carried out at every time step. Obviously this is unnecessary but it has been done to check whether the algorithm triggers possible instabilities. Fig. 8 illustrates the deformed shape of a bending column where colour indicates pressure contour. Remarkably, even with a small number of particles (M1), the obtained result agrees very well with the Total Lagrangian result. No instabilities are observed. As shown in Fig. 9a, the amount of numerical dissipation reduces with an increasing number of particles. For completeness, we also simulate the same problem by updating the reference configuration at every 3, 30 and 300 time steps and the same results are obtained (see Fig. 9b).



Fig. 5. Tensile mode. Pressure distribution for model M4 at two different time instants, namely t = 0.215 s (first row) and t = 0.3 s (second row). Results are obtained using the four options for the computation of the kernel function and its gradient presented in Section 4.2. Update of the reference domain is carried out at every time step of the time integration process.

#### 8.1.3. Scenario III : Twisting mode

In order to examine the robustness of the algorithm, a well documented twisting example introduced in [24–26,71] is considered. Fig. 10 compares the deformation process of the structure at time t = 0.1 s using the four model refinements previously described. Similar results in terms of deformed shape and pressure field are observed. Finally, and for qualitative comparison, Fig. 11 monitors the evolution of the accumulated angle at four different positions,<sup>3</sup> namely  $N_{11}$ ,  $N_{12}$ ,  $N_{13}$  and  $N_{14}$ . Comparing the proposed algorithm with the published Total Lagrangian SPH algorithm [26], no noticeable difference is observed.

<sup>&</sup>lt;sup>3</sup> Notice that the initial angles at positions  $\{N_{11}, N_{12}, N_{13}, N_{14}\}$  are measured from the axis X in a counter-clockwise direction. For instance, the initial angles of  $\{N_{11}, N_{12}, N_{13}, N_{14}\}$  are  $\theta = \{225^\circ, 315^\circ, 45^\circ, 135^\circ\}$ , respectively.



Fig. 6. Tensile mode. Evolution of total numerical dissipation for model M1 considering different options of kernel evaluations. Results are obtained by updating the reference domain at every time step of the time integration process.

Table 1	2
---------	---

L-shaped structure: material and algorithmic parameters used in the simulation.

Material density	$\rho_R$	1100	$\rm kg \ m^{-3}$
Young's modulus	Ε	50,500	$N m^{-2}$
Poisson's ratio	ν	0.3	
Kernel type	Quadratic kernel function		
SPH correction	Option#2		
Smoothing factor	$f_h$	0.6	
CFL	$\alpha_{\rm CFL}$	0.9	

#### 8.2. L-shaped structure

As previously explored in Refs. [25,59,66,72,73], the main objective of this classical benchmark problem is to examine the capability of the proposed algorithm of preserving both the linear and angular momenta of a system. An L-shaped structure (see Fig. 12) is subjected to an external torque induced by a pair of time-varying tractions acting on two of its boundary faces  $F_1(t)$  and  $F_2(t)$ , described as

$$\boldsymbol{F}_{1}(t) = -\boldsymbol{F}_{2}(t) = \begin{bmatrix} 150\\ 300\\ 450 \end{bmatrix} f(t) \text{ (N/m}^{2}), \qquad f(t) = \begin{cases} t & 0 \le t < 2.5 \text{ s}, \\ 5 - t & 2.5 \le t < 5 \text{ s}, \\ 0 & t \ge 5 \text{ s}. \end{cases}$$
(68)

In this example, a neo-Hookean model is considered. The values of all the parameters and the associated SPH ingredients used in the simulation can be found in Table 2. For completeness, three different levels of particle refinements are considered. More precisely, {**M1**, **M2**, **M3**} comprise {828, 5445, 13950} number of SPH particles, respectively. To assign the volume of each SPH particle, a tributary volume distribution is used.

First, a particle refinement study for the proposed SPH algorithm is carried out. In Fig. 13, the deformation pattern of the structure predicted using a small number of particles (M1) agrees extremely well with the results obtained using finer discretisations (M2 and M3). For comparison purpose, a Total Lagrangian Upwind SPH [26] discretised with M3 is also included and compared. Practically identical pressure profiles are observed (see Fig. 13).

Second, Figs. 14a and 14b examine the ability of the proposed algorithm (discretised with M3) of preserving both the linear momentum and angular momentum of the system. Specifically, the global linear momentum is close to



Fig. 7. Tensile mode. Evolution of total numerical dissipation for different model refinements using (a) **Option#1**, (b) **Option#2**, (c) **Option#3** and (d) **Option#4**. Results are obtained by updating the reference domain at every time step of the time integration process.

(and oscillates around) zero machine accuracy at all times as no movement of the centre of mass is appreciated. The global angular momentum is however expected to be conserved after the loading phase, that is when time t > 5 s. Additionally, Fig. 14c illustrates the time histories of different forms of energy obtained using the proposed algorithm and the Total Lagrangian SPH. These include kinetic energy, strain energy and total energy, where the latter is defined as the sum of kinetic and strain energies. Identical results are observed. Given the fact that some consistent upwinding stabilisation terms are introduced in our SPH algorithm, a slight decrease in the total energy



Fig. 8. Bending mode. Pressure distribution at time t = 0.45 s using four different model refinements, namely (M1)  $5 \times 5 \times 25$ , (M2)  $7 \times 7 \times 35$ , (M3)  $9 \times 9 \times 49$  and (M4)  $17 \times 17 \times 97$  number of particles. First four columns show the results obtained using the proposed algorithm with updates performed at every time step of the time integration process. The last column shows Total Lagrangian SPH results for comparison purposes. (For interpretation of the references to colour in this figure legend, the reader is referred to the web version of this article.)



Fig. 9. Bending mode. Evolution of total numerical dissipation for (a) four different model refinements with updates performed at every time step of the time integration process, and (b) M1 model with different frequent updates of the referential domain.

is unavoidable after the loading phase. As expected, numerical dissipation decreases with a progressive level of refinement, as reported in Fig. 14d.



-2.00e + 06 - 1.56e + 06 - 1.11e + 06 - 6.67e + 05 - 2.22e + 05 - 2.22e + 05 - 6.67e + 05 - 1.11e + 06 - 1.56e + 06 - 2.00e + 06 - 0.00e + 0.00e +

Fig. 10. Twisting mode. Pressure distribution at time t = 0.1 s using four different model refinements, namely (M1)  $5 \times 5 \times 25$ , (M2)  $7 \times 7 \times 35$ , (M3)  $9 \times 9 \times 49$  and (M4)  $17 \times 17 \times 97$  number of particles. First four columns show the results obtained using the proposed algorithm with updates performed at every time step of the time integration process. The last column shows Total Lagrangian SPH results for comparison purposes.

Table	3
-------	---

Impact bar: material and data simulation used.

Material density	$\rho_R$	8930	$\rm kg \ m^{-3}$
Young's modulus	Ε	117	GPa
Poisson's ratio	ν	0.35	
Yield stress	$\tau_{v}^{0}$	0.4	GPa
Linear hardening modulus	Ĥ	0.1	GPa
Kernel type	Quadratic kernel function		
Smoothing factor	$f_h$	0.6	
Gradient correction	Option#1		
CFL	$\alpha_{\rm CFL}$	0.9	

#### 8.3. Impact bar

In this example, a copper bar of length L = 0.0324 m and of radius R = 0.0032 m impacts against a rigid wall with a dropping velocity given by  $v_0 = [0, 0, -V_0]^T$ , where  $V_0 = 227$  m s<sup>-1</sup>. The main purpose of the example is to assess the performance of the proposed algorithm in capturing large plastic flows under high speed impact [26,74]. The geometry of the problem is illustrated in Fig. 15. A (Hencky-based) von Mises hyperelastic–plastic material, in conjunction with isotropic linear hardening rule, is chosen for the simulation of this problem. A summary of data simulation and material properties used are reported in Table 3. To assign the volume for every particle, a tributary volume is used.

Due to the symmetry of the problem, only a quarter of the geometry is represented and discretised using three levels of refinement, namely (M1) 1560, (M2) 3744 and (M3) 7280 particles. Fig. 16 compares the deformation process of the structure at time  $t = \{40, 60, 80\}$  µs using these three models. Similar deformed shapes are observed, with colour contours indicating equivalent von Mises stress and plastic strain profiles. In terms of the energy plot (see Fig. 17a), the kinetic energy of the system decreases upon impact. Such energy first transforms into elastic strain energy (prior to plasticity), followed by the addition of plastic dissipation (within the plastic regime). Moreover, given the nature of the proposed algorithm, a very small amount of the kinetic energy also converts to monotonic decreasing numerical dissipation during the deformation process, as displayed in Fig. 17b.



Fig. 11. Twisting mode. Time history of twisting angle for model M4 at four different locations, namely (a)  $N_{14}$ , (b)  $N_{13}$ , (c)  $N_{11}$  and (d)  $N_{12}$ .



Fig. 13. L-shaped block. Pressure distribution at time t = 7.8 s using three different model refinements, namely (M1) 828, (M2) 5445 and (M3) 13950 number of particles. First three columns show the results obtained using the proposed SPH algorithm with updates performed at every time step of the time integration process. The last column shows Total Lagrangian SPH results for comparison purposes.

For verification purposes, it is worth noticing that the results obtained using the proposed SPH algorithm match extremely well with those of the Total Lagrangian SPH counterpart. The final radius of the copper bar at time  $t = 80 \,\mu\text{s}$  predicted by the proposed algorithm is shown in Fig. 18b, benchmarked against other published numerical results [26,75].<sup>4</sup> As shown in Refs. [75], the solutions obtained using the standard linear 4-noded tetrahedra (being widely used in commercial software) typically suffers from volumetric locking and pressure instabilities. The proposed meshfree method clearly circumvents these issues.

#### 8.4. Necking bar

The following benchmark example [76–80] illustrates necking of a cylindrical bar, of radius R = 0.006413 m and length L = 0.053334 m, under tension. The complete geometry of the bar can be viewed in Fig. 19. Notice that a geometric imperfection (one percent reduction in the radius) is introduced to induce necking in the central region

<sup>&</sup>lt;sup>4</sup> For the same level of accuracy, and due to the higher numerical dissipation of the SPH method, the results obtained with the SPH algorithm have been obtained with slightly finer discretisation than those of the robust Petrov–Galerkin Finite Element Method explored by the authors in their previous publications [48-50,58,59]. The latter however requires a few user-defined stabilisation parameters.



Fig. 14. L-shaped block. Evolution of (a) the components of global linear momentum, (b) the components of global angular momentum and (c) different forms of energy (e.g. kinetic energy K, strain energy  $\Psi$  and total energy  $E = K + \Psi$ ) using M3 model. In addition, the time evolution of (d) global numerical dissipation for three different model refinements is also monitored. Notice that  $[\bullet]_{ULF}$  refers to the results obtained via the proposed Updated Reference Lagrangian SPH algorithm and  $[\bullet]_{TLF}$  refers to the results from the published Total Lagrangian SPH algorithm [26].



Fig. 15. Impact bar.

Table 4						
Necking	bar:	material	and	data	simulation	used.

Material density	ρ <sub>R</sub>	7850	kg m <sup>-3</sup>
Young's modulus	E	206.9	GPa
Poisson's ratio	ν	0.29	
Yield stress	$\tau_v^0$	0.45	GPa
Residual yield stress	$\tau_{v}^{\infty}$	0.715	GPa
Linear hardening modulus	Ή.	129.24	MPa
Saturation exponent	δ	16.93	
Kernel type	Quadratic kernel function		
Smoothing factor	$f_h$	0.6	
Gradient correction	Option#1		
CFL	$\alpha_{\rm CFL}$	0.9	

of the bar. The top surface of the cylindrical bar is stretched with smooth time-varying velocity profile described as

$$\boldsymbol{v}_{0} = \begin{bmatrix} 0\\0\\V \end{bmatrix}; \qquad V = \begin{cases} V_{0} + (V_{1} - V_{0})\,\xi_{01}^{3}\left(10 - 15\xi_{01} + 6\xi_{01}^{2}\right); & \text{if} & 0 \le t \le t_{1} \\ V_{1} + (V_{2} - V_{1})\,\xi_{12}^{3}\left(10 - 15\xi_{12} + 6\xi_{12}^{2}\right); & \text{if} & t_{1} < t \le t_{2} \end{cases}, \tag{69}$$

with

$$\xi_{01} = \frac{t - t_0}{t_1 - t_0}; \qquad \xi_{12} = \frac{t - t_1}{t_2 - t_1}.$$
(70)

Here, the value of the parameters are  $t_0 = V_0 = V_2 = 0$ ,  $t_1 = 0.0007$  s,  $V_1 = 10 \text{ ms}^{-1}$  and  $t_2 = 0.0014$  s. Graphical representation of this bell-shaped velocity profile (69) is depicted in Fig. 19c.

The material properties of the (Hencky-based) von Mises material and the simulation parameters are summarised in Table 4. Nonlinear hardening rule as proposed in [76] is chosen.

Due to the presence of symmetry planes, only a slice (of  $\frac{\pi}{16}$  rad) of the upper half of the cylinder is simulated with appropriate boundary conditions. For completeness, two levels of particle refinement for the model are studied. Model **M1** contains a number of 1428 SPH particles and model **M2** comprises 5535 particles. The particle distributions are non-uniform and each particle carries a volume tributarily distributed from the associated



Fig. 16. Impact bar. A sequence of deformed structures at time  $t = \{40, 60, 80\} \ \mu s$  (from top to bottom) using  $\{M1, M2, M3\}$  (left to right). In terms of contour plot, von Mises stresses (left side) and equivalent plastic strain (right side). (For interpretation of the references to colour in this figure legend, the reader is referred to the web version of this article.)



Fig. 17. Impact bar. Time history of (a) different forms of energy (e.g. kinetic energy K, the sum of strain energy  $\Psi$  and plastic dissipation  $W_p$ , and total energy being defined as  $K + \Psi + W_p$ ) using M3 and (b) global numerical dissipation for three model refinements. Notice that  $[\bullet]_{ULF}$  refers to the results obtained via the proposed Updated Reference Lagrangian SPH algorithm and  $[\bullet]_{TLF}$  refers to the results from the published Total Lagrangian SPH algorithm [26].



**Fig. 18.** Impact bar. (a) Pressure comparison at 80  $\mu$ s for **M3**. Results obtained via Total Lagrangian SPH (left side) and the proposed SPH algorithm (right side), and (b) evolution of the radius of the bar measured at the impact interface for three different model refinements. Notice that  $[\bullet]_{ULF}$  refers to the results obtained via the proposed Updated Reference Lagrangian SPH algorithm and  $[\bullet]_{TLF}$  refers to the results from the published Total Lagrangian SPH algorithm [26].

P.R.R. de Campos, A.J. Gil, C.H. Lee et al.

Computer Methods in Applied Mechanics and Engineering 392 (2022) 114680



Fig. 19. (a) Problem set-up; (b) Necking bar details and (c) Applied bell-shaped velocity profile (69).



Fig. 20. Non-uniform distribution of SPH particles for M1 and M2.

neighbouring particles. In order to accurately capture the onset of necking, more particles are placed within the crucial necking zone, see Fig. 20.

Fig. 21 shows the time evolution of the deformation pattern with both plastic strain and equivalent von Mises stress plotted. The proposed method can clearly capture the formation of necking at the central region of the bar. Additionally, Fig. 22a depicts a comparison of the proposed SPH algorithm against the recently published Total Lagrangian SPH formulation [26] at the end of the simulation (that is, when the total elongation of the bar is of 14 mm). Remarkably, both formulations produce very similar results in terms of deformed shape and pressure field. For qualitative comparison, we also monitor the ratio between current and initial radius of the bar in necking regime as a function of the elongation. Indeed, our results agree extremely well with those of published experimental and numerical results [79,80]. This can be seen in Fig. 22b.

For visualisation purposes, Fig. 23 displays a three dimensional view of the cylindrical bar at different levels of deformation. No spurious hourglassing is observed in the necking zone.



Fig. 21. Necking bar. A sequence of deformed structures when the total elongation of the bar is of  $\{10, 12, 14\}$  mm (from top to bottom) using  $\{M1, M2\}$  (left to right). In terms of contour plot, von Mises stress profile (left side) and equivalent plastic strain (right side).



Fig. 22. Necking bar. (a) Pressure comparison when the total elongation of the bar is 14 mm for model M2. Results obtained via Total Lagrangian SPH (left side) and the proposed SPH algorithm (right side), and (b) radius reduction in the necking region compared against published results [76]. Notice that  $[\bullet]_{ULF}$  refers to the results obtained via the proposed Updated Reference Lagrangian SPH algorithm and  $[\bullet]_{TLF}$  refers to the results from the published Total Lagrangian SPH algorithm [26].

#### 9. Conclusions

In this paper, a new Updated Reference Lagrangian Smooth Particle Hydrodynamics algorithm for the analysis of large deformation isothermal elasticity and elasto-plasticity has been presented. From the continuum point of view, by exploiting as referential configuration an intermediate configuration of the deformation process, the equation of conservation of linear momentum and three geometric conservation laws (for the deformation gradient, its cofactor and its determinant) have been rewritten leading to a very generic (incremental) system of first order conservation laws. Moreover, this can be degenerated into a Total Lagrangian system (previously pursued by the authors in [24–26]) or into a purely Updated Lagrangian system. Technically speaking, only the geometric conservation law for the deformation gradient (out of the three geometric conservation laws) is strictly necessary. An appropriate multiplicative decomposition of the conventional Total Lagrangian system, apart from the use of alternative referential (incremental) conservation variables and corresponding entropy (work) conjugate fields.

From the computational implementation standpoint, a second order (in space and time) entropy-stable SPH upwinding stabilisation method is employed by means of the use of the Rankine Hugoniot jump conditions, without resorting to any ad-hoc algorithmic regularisation. A new three-stage Runge–Kutta time integration method is implemented resulting in values of the CFL stability restriction close to one. With the aim of demonstrating the robustness and potential of the methodology, an ample suite of challenging problems (in elasticity and elastoplasticity) have been presented and thoroughly analysed. The amount of numerical dissipation introduced by the algorithm has been strictly monitored through the extra implementation of a conservation equation for the total energy density, though not strictly required from the modelling point of view. The use of alternative kernels and gradients of kernels as well the use of different strategies to select the referential (incremental) configuration has been explored.

The next step of our work is the adaptation of the current Updated Reference Lagrangian SPH framework to handle dynamic fracture problems typically accompanied by complex topological changes caused by material separation.



**Fig. 23.** Necking bar. A sequence of deformed structures with von Mises profile when the total elongation of the bar is of  $\{0.028, 0.054, 0.09, 0.124, 1.854, 5.864, 9.852, 10.956, 11.642, 12.342, 12.704, 14\}$  mm (from left to right and top to bottom). Results obtained via model **M2** with updates performed at every time step of the time integration process.

# **Declaration of competing interest**

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

## Acknowledgements

The authors would like to acknowledge the financial support received through the project Marie Skłodowska-Curie ITN-EJD ProTechTion, funded by the European Union Horizon 2020 research and innovation program with Grant Number 764636. The third author also gratefully acknowledges the support provided by the EPSRC Strategic Support Package: Engineering of Active Materials by Multiscale/Multiphysics Computational Mechanics - EP/R008531/1. The fourth author acknowledges the support of the Serra Húnter Programme of the Generalitat de Catalunya, the Spanish Ministry of Economy and Competitiveness (Grant agreement No. PID2020-113463RB-C33) and the Spanish Ministry of Economy and Competitiveness through the Severo Ochoa programme for centres of excellence in RTD (Grant agreement No. CEX2018-000797-S).

#### References

- T. Belytschko, Y. Krongauz, D. Organ, M. Fleming, P. Krysl, Meshless methods: An overview and recent developments, Comput. Methods Appl. Mech. Engrg. 139 (1-4) (1996) 3–47, http://dx.doi.org/10.1016/S0045-7825(96)01078-X.
- [2] A. Huerta, T. Belytschko, S. Fernández-Méndez, T. Rabczuk, X. Zhuang, M. Arroyo, Meshfree methods, Encycl. Comput. Mech. Second Ed. 1 (1) (2017) 1–38, http://dx.doi.org/10.1002/9781119176817.ecm2005.
- [3] J.-S. Chen, M. Hillman, S.-W. Chi, Meshfree methods: Progress made after 20 years, J. Eng. Mech. 143 (4) (2017) 1–38, http://dx.doi.org/10.1061/(ASCE)EM.1943-7889.0001176.
- [4] L.W. Zhang, A.S. Ademiloye, K.M. Liew, Meshfree and particle methods in biomechanics: Prospects and challenges, Arch. Comput. Methods Eng. 26 (2019) 1547–1576, http://dx.doi.org/10.1007/s11831-018-9283-2.
- [5] T. Belytschko, T. Rabczuk, A. Huerta, S. Fernández-Méndez, Meshfree methods, in: Encyclopedia of Computational Mechanics, American Cancer Society, 2004.
- [6] R.A. Gingold, J.J. Monaghan, Smoothed particle hydrodynamics: theory and application to non-spherical stars, Mon. Not. R. Astron. Soc. 181 (3) (1977) 375–389, http://dx.doi.org/10.1093/mnras/181.3.375.
- [7] L.B. Lucy, A numerical approach to the testing of the fission hypothesis, Astron. J. 82 (12) (1977) 1013–1024.
- [8] L.D. Libersky, A.G. Petschek, Smooth particle hydrodynamics with strength of materials, Lect. Notes Phys. 395 (1) (1991) 248–257, http://dx.doi.org/10.1007/3-540-54960-9\_58.
- [9] L.D. Libersky, A.G. Petschek, T.C. Carney, J.R. Hipp, F.A. Allahdadi, High strain Lagrangian hydrodanymics: A three-dimensional SPH code for dynamic material response, J. Comput. Phys. 109 (1) (1993) 65–75, http://dx.doi.org/10.1006/jcph.1993.1199.
- [10] J.J. Monaghan, Smoothed particle hydrodynamics, Annu. Rev. Astron. Astrophys. 30 (1) (1992) 543–574.
- [11] J.W. Swegle, S.W. Attaway, M.W. Heinstein, F.J. Mello, D.L. Hicks, An Analysis of Smoothed Particle Hydrodynamics, Tech. Rep. SAND93-2513, Sandia National Laboratories, 1994.
- [12] R. Vignjevic, J. Campbell, Review of development of the smooth particle hydrodynamics (SPH) method, in: S. Hiermaier (Ed.), Predict. Model. Dyn. Process. 1 (1) (2009) 367–396, http://dx.doi.org/10.1007/978-1-4419-0727-1\_20.
- [13] J.W. Swegle, D.L. Hicks, S.W. Attaway, Smoothed particle hydrodynamics stability analysis, J. Comput. Phys. 116 (1) (1995) 123-134.
- [14] C.T. Dyka, R.P. Ingel, Addressing Tension Instability in SPH Methods, Tech. Rep. NRL/MR/6384–94–7641, Naval Research Laboratory, 1994.
- [15] C.T. Dyka, R.P. Ingel, An approach for tension instability in smoothed particle hydrodynamics (SPH), Comput. Struct. 57 (4) (1995) 573–580, http://dx.doi.org/10.1016/0045-7949(95)00059-P.
- [16] J.P. Morris, A study of the stability properties of smooth particle hydrodynamics, Publ. Astron. Soc. Aust. 13 (1) (1996) 97–102, http://dx.doi.org/10.1017/S1323358000020610.
- [17] P.W. Randles, L.D. Libersky, Smoothed particle hydrodynamics: Some recent improvements and applications, Comput. Methods Appl. Mech. Engrg. 139 (1) (1996) 375–408, http://dx.doi.org/10.1016/S0045-7825(96)01090-0.
- [18] G.A. Dilts, Moving least-squares particle hydrodynamics I: Consistency and stability, Internat. J. Numer. Methods Engrg. 44 (1) (1999) 1115–1155, http://dx.doi.org/10.1002/(SICI)1097-0207(19990320)44:8<1115::AID-NME547>3.0.CO;2-L.
- [19] T. Belytschko, Y. Guo, W.K. Liu, S.P. Xiao, A unified stability analysis of meshless particle methods, Internat. J. Numer. Methods Engrg. 1 (48) (2000) 1359–1400, http://dx.doi.org/10.1002/1097-0207(20000730)48:9<1359::AID-NME829>3.0.CO;2-U.
- [20] J. Bonet, S. Kulasegaram, Correction and stabilisation of smooth particle hydrodynamics methods with applications in metal forming simulations, Internat. J. Numer. Methods Engrg. 47 (1) (2000) 1189–1214, http://dx.doi.org/10.1002/(SICI)1097-0207(20000228)47: 6<1189::AID-NME830>3.0.CO;2-I.
- [21] R. Vignjevic, J. Campbell, L. Libersky, A treatment of zero-energy modes in the smoothed particle hydrodynamics method, Comput. Methods Appl. Mech. Engrg. 184 (1) (2000) 67–85, http://dx.doi.org/10.1016/S0045-7825(99)00441-7.
- [22] J. Bonet, S. Kulasegaram, Remarks on tension instability of Eulerian and Lagrangian corrected smooth particle hydrodynamics (CSPH) methods, Internat. J. Numer. Methods Engrg. 52 (1) (2001) 1203–1220, http://dx.doi.org/10.1002/nme.242.
- [23] Y. Vidal, J. Bonet, A. Huerta, Stabilized updated Lagrangian corrected SPH for explicit dynamics problems, Internat. J. Numer. Methods Engrg. 69 (1) (2007) 2687–2710, http://dx.doi.org/10.1002/nme.1859.
- [24] C.H. Lee, A.J. Gil, G. Greto, S. Kulasegaram, J. Bonet, A new Jameson-Schmidt-turkel smooth particle hydrodynamics algorithm for large strain explicit fast dynamics, Comput. Methods Appl. Mech. Engrg. 311 (1) (2016) 71–111, http://dx.doi.org/10.1016/j.cma.2016. 07.033.

- [25] C.H. Lee, A.J. Gil, O.I. Hassan, J. Bonet, S. Kulasegaram, A variationally consistent streamline upwind Petrov-Galerkin smooth particle hydrodynamics algorithm for large strain solid dynamics, Comput. Methods Appl. Mech. Engrg. 318 (1) (2017) 514–536, http://dx.doi.org/10.1016/j.cma.2017.02.002.
- [26] C.H. Lee, A.J. Gil, A. Ghavamian, J. Bonet, A total Lagrangian upwind smooth particle hydrodynamics algorithm for large strain explicit solid dynamics, Comput. Methods Appl. Mech. Engrg. 344 (1) (2019) 209–250, http://dx.doi.org/10.1016/j.cma.2018.09.033.
- [27] Y. Krongauz, T. Belytschko, Consistent pseudo-derivatives in meshless methods, Comput. Methods Appl. Mech. Engrg. 146 (1) (1997) 371–386, http://dx.doi.org/10.1016/S0045-7825(96)01234-0.
- [28] J. Bonet, T.S.L. Lok, Variational and momentum preservation aspects of smooth particle hydrodynamics formulations, Comput. Methods Appl. Mech. Engrg. 180 (1) (1999) 97–115, http://dx.doi.org/10.1016/S0045-7825(99)00051-1.
- [29] G.A. Dilts, Moving least-squares particle hydrodynamics II: Conservation and boundaries, Internat. J. Numer. Methods Engrg. 48 (1) (2000) 1503–1524, http://dx.doi.org/10.1002/1097-0207(20000810)48:10<1503::AID-NME832>3.0.CO;2-D.
- [30] W.G. Hoover, C.G. Hoover, E.C. Merritt, Smooth-particle applied mechanics: Conservation of angular momentum with tensile stability and velocity averaging, Phys. Rev. 69 (1) (2004) 016702, http://dx.doi.org/10.1103/PhysRevE.69.016702.
- [31] G.C. Ganzenmüller, An hourglass control algorithm for Lagrangian smooth particle hydrodynamics, Comput. Methods Appl. Mech. Engrg. 286 (2015) 87–106.
- [32] G.C. Ganzenmüller, S. Hiermaier, M. May, On the similarity of meshless discretizations of peridynamics and smooth-particle hydrodynamics, Comput. Struct. 150 (2015) 71–78.
- [33] H. Ren, X. Zhuang, Y. Cai, T. Rabczuk, Dual-horizon perydynamics, Int. J. Numer. Methods Eng. 108 (2016) 1451–1476.
- [34] Z. Dai, H. Ren, Z. Zhuang, T. Rabczuk, Dual-support smoothed particle hydrodynamics for elastic mechanics, Int. J. Comput. Methods 14 (2017) 1750039.
- [35] C. Zhang, J. Wang, M. Rezavand, D. Wu, X. Hu, An integrative smoothed particle hydrodynamics method for modeling cardiac function, Comput. Methods Appl. Mech. Engrg. 381 (2021) 113847, http://dx.doi.org/10.1016/j.cma.2021.113847.
- [36] C. Zhang, M. Rezavand, Y. Zhu, Y. Yu, D. Wu, W. Zhang, S. Zhang, J. Wang, X. Hu, Sphinxsys: An open-source meshless, multi-resolution and multi-physics library, Softw. Impacts 6 (2020) 100033, http://dx.doi.org/10.1016/j.simpa.2020.100033.
- [37] A. Khayyer, H. Gotoh, Y. Shimizu, Y. Nishijima, 3D Lagrangian meshfree projection-based solver for hydroelastic fluid-structure interactions, J. Fluids Struct. 105 (2021) 103342, http://dx.doi.org/10.1016/j.jfluidstructs.2021.103342.
- [38] Y. You, A. Khayyer, X. Zheng, H. Gotoh, Q. Ma, Enhancement of δ-SPH for ocean engineering applications through incorporation of a background mesh scheme, Appl. Ocean Res. 110 (2021) 102508, http://dx.doi.org/10.1016/j.apor.2020.102508.
- [39] A. Khayyer, H. Gotoh, Enhancement of stability and accuracy of the moving particle semi-implicit method, J. Comput. Phys. 230 (2011) 3093–3118, http://dx.doi.org/10.1016/j.jcp.2011.01.009.
- [40] C.G. Koh, M. Luo, W. Bai, Modelling of liquid sloshing with constrained floating baffle, Comput. Struct. 122 (2013) 270–279, http://dx.doi.org/10.1016/j.compstruc.2013.03.018.
- [41] M. Luo, A. Khayyer, P. Lin, Particle methods in ocean and coastal engineering, Appl. Ocean Res. 114 (2021) 102734, http: //dx.doi.org/10.1016/j.apor.2021.102734.
- [42] C. Zhang, M. Rezavand, X. Hu, A multi-resolution SPH method for fluid-structure interactions, J. Comput. Phys. 429 (2021) 110028, http://dx.doi.org/10.1016/j.jcp.2020.110028G.
- [43] D.A. Abdoh, A.S. Ademiloye, K.M. Liew, Modeling glass cooling mechanism with down-flowing water film via the smoothed particle hydrodynamics, Comput. Methods Appl. Mech. Engrg. 362 (2020) 112839, http://dx.doi.org/10.1016/j.cma.2020.112839.
- [44] A.S. Ademiloye, L.W. Zhang, K.M. Liew, A multiscale framework for large deformation modeling of rbc membranes, Comput. Methods Appl. Mech. Engrg. 329 (2018) 144–167, http://dx.doi.org/10.1016/j.cma.2017.10.004.
- [45] G.C. Ganzenmüller, M. Sauer, M. May, S. Hiermaier, Hourglass control for smooth particle hydrodynamics removes tensile and rank-deficiency instabilities, Eur. Phys. J. Spec. Top. 225 (2016) 385–395, http://dx.doi.org/10.1140/epjst/e2016-02631-x.
- [46] S. Leroch, M. Varga, S.J. Eder, A. Vernes, M.R. Ripoll, G.C. Ganzenmüller, Smooth particle hydrodynamics simulation of damaged induced by a spherical indenter scratching a viscoplastic material, Int. J. Solids Struct. 81 (2016) 188–202, http://dx.doi.org/10.1016/j. ijsolstr.2015.11.025.
- [47] C.H. Lee, A.J. Gil, J. Bonet, Development of a cell centred upwind finite volume algorithm for a new conservation law formulation in structural dynamics, Comput. Struct. 118 (1) (2013) 13–38, http://dx.doi.org/10.1016/j.compstruc.2012.12.008.
- [48] C.H. Lee, A.J. Gil, J. Bonet, Development of a stabilised Petrov–Galerkin formulation for conservation laws in Lagrangian fast solid dynamics, Comput. Methods Appl. Mech. Engrg. 268 (1) (2014) 40–64, http://dx.doi.org/10.1016/j.cma.2013.09.004.
- [49] A.J. Gil, C.H. Lee, J. Bonet, M. Aguirre, A stabilised Petrov–Galerkin formulation for linear tetrahedral elements in compressible, nearly incompressible and truly incompressible fast dynamics, Comput. Methods Appl. Mech. Engrg. 276 (1) (2014) 659–690, http://dx.doi.org/10.1016/j.cma.2014.04.006.
- [50] J. Bonet, A.J. Gil, C.H. Lee, M. Aguirre, R. Ortigosa, A first order hyperbolic framework for large strain computational solid dynamics. Part I: Total Lagrangian isothermal elasticity, Comput. Methods Appl. Mech. Engrg. 283 (1) (2015) 689–732, http: //dx.doi.org/10.1016/j.cma.2014.09.024.
- [51] J. Bonet, A.J. Gil, R. Ortigosa, A computational framework for polyconvex large strain elasticity, Comput. Methods Appl. Mech. Engrg. 283 (1) (2015) 1061–1094, http://dx.doi.org/10.1016/j.cma.2014.10.002.
- [52] J. Bonet, A.J. Gil, R. Ortigosa, On a tensor cross product based formulation of large strain solid mechanics, Int. J. Solids Struct. 84 (1) (2016) 49–63, http://dx.doi.org/10.1016/j.ijsolstr.2015.12.030.
- [53] J. Bonet, A.J. Gil, Mathematical models of supersonic and intersonic crack propagation in linear elastodynamics, Int. J. Fract. 229 (2021) 55–75, http://dx.doi.org/10.1007/s10704-021-00541-y.

- [54] J. Haider, C.H. Lee, A.J. Gil, A. Huerta, J. Bonet, An upwind cell centred total Lagrangian finite volume algorithm for nearly incompressible explicit fast solid dynamic applications, Comput. Methods Appl. Mech. Engrg. 1 (340) (2018) 684–727, http: //dx.doi.org/10.1016/j.cma.2018.06.010.
- [55] O.I. Hassan, A. Ghavamian, C.H. Lee, A.J. Gil, J. Bonet, F. Auricchio, An upwind vertex centred finite volume algorithm for nearly and truly incompressible explicit fast solid dynamics applications: Total and updated Lagrangian formulations, J. Comput. Phys.: X 3 (2019) 100025, http://dx.doi.org/10.1016/j.jcpx.2019.100025.
- [56] I.A. Karim, C.H. Lee, A.J. Gil, J. Bonet, A two-steo taylor-Galerkin formulation for fast dynamics, Eng. Comput. 31 (2014) 366–387, http://dx.doi.org/10.1108/EC-12-2012-0319.
- [57] M. Aguirre, A.J. Gil, J. Bonet, C.H. Lee, An upwind vertex centred finite volume solver for Lagrangian solid dynamics, J. Comput. Phys. 300 (2015) 387–422, http://dx.doi.org/10.1016/j.jcp.2015.07.029.
- [58] A.J. Gil, C.H. Lee, J. Bonet, R. Ortigosa, A first order hyperbolic framework for large strain computational solid dynamics. Part II: Total Lagrangian compressible, nearly incompressible and truly incompressible elasticity, Comput. Methods Appl. Mech. Engrg. 300 (1) (2016) 146–181, http://dx.doi.org/10.1016/j.cma.2015.11.010.
- [59] J. Bonet, C. Lee, A. Gil, A. Ghavamian, A first order hyperbolic framework for large strain computational solid dynamics. Part III: Thermo-elasticity, Comput. Methods Appl. Mech. Engrg. 373 (2021) http://dx.doi.org/10.1016/j.cma.2020.113505.
- [60] J. Bonet, A.J. Gil, R.D. Wood, Nonlinear Solid Mechanics for Finite Element Analysis: Dynamics, Cambridge University Press, 2020.
- [61] K.W.Q. Low, C.H. Lee, A.J. Gil, J. Haider, J. Bonet, A parameter-free total Lagrangian smooth particle hydrodynamics algorithm applied to problems with free surfaces, Comput. Part. Mech. 8 (2021) 859–892, http://dx.doi.org/10.1007/s40571-020-00374-x.
- [62] A. Khayyer, Y. Shimizu, H. Gotoh, K. Nagashima, A coupled incompressible SPH-Hamiltonian SPH solver for hydroelastic FSI corresponding to composite structures, Appl. Math. Model. 94 (2021) 242–271, http://dx.doi.org/10.1016/j.apm.2021.01.011.
- [63] J. Bonet, M.X. Rodríguez-Paz, Hamiltonian formulation of the variable-h SPH equations, J. Comput. Phys. 209 (2) (2005) 541–558, http://dx.doi.org/10.1016/j.jcp.2005.03.030.
- [64] G.R. Johnson, R.A. Stryk, S.R. Beissel, SPH for high velocity impact computations, Comput. Methods Appl. Mech. Engrg. 139 (1) (1996) 347–373, http://dx.doi.org/10.1016/S0045-7825(96)01089-4.
- [65] J.M. Owen, J.V. Villumsen, P.R. Shapiro, H. Martel, Adaptive smoothed particle hydrodynamics: Methodology II, Astrophys. J. Suppl. Ser. 116 (2) (1998) 155–209, http://dx.doi.org/10.1002/nme.684.
- [66] A. Ghavamian, A.J. Gil, C.H. Lee, J. Bonet, T. Heuze, L. Stainier, An entropy stable smooth particle hydrodynamics algorithm for large strain thermo-elasticity, Comput. Methods Appl. Mech. Engrg. 379 (1) (2021) 113736, http://dx.doi.org/10.1016/j.cma.2021.113736.
- [67] C.-W. Shu, S. Osher, Efficient implementation of essentially non-oscillatory shock-capturing schemes, J. Comput. Phys. 1 (77) (1988) 439–471, http://dx.doi.org/10.1016/0021-9991(88)90177-5.
- [68] J. Haider, C.H. Lee, A.J. Gil, J. Bonet, A first-order hyperbolic framework for large strain computational solid dynamics: An upwind cell centred total Lagrangian scheme, Internat. J. Numer. Methods Engrg. 109 (3) (2016) 407–456, http://dx.doi.org/10.1002/nme.5293.
- [69] G. Scovazzi, B. Carnes, X. Zeng, S. Rossi, A simple, stable, and accurate linear tetrahedral finite element for transient, nearly and fully incompressible solid dynamics: A dynamic variational multiscale approach, Internat. J. Numer. Methods Engrg. 106 (2016) 799–839.
- [70] S. Rossi, N. Abboud, G. Scovazzi, Implicit finite incompressible elastodynamics with linear finite elements: A stabilized method in rate form, Comput. Methods Appl. Mech. Engrg. 311 (2016) 208–249.
- [71] G. Scovazzi, T. Song, X. Zeng, A velocity/stress mixed stabilized nodal finite element for elastodynamics: Analysis and computations with strongly and weakly enforced boundary conditions, Comput. Methods Appl. Mech. Engrg. 325 (2017) 532–576, http://dx.doi.org/ 10.1016/j.cma.2017.07.018.
- [72] M. Franke, A. Janz, M. Schiebl, P. Betsch, An energy momentum consistent integration scheme using a polyconvexity-based framework for nonlinear thermo-elastodynamics, Int. J. Comput. Methods Eng. Sci. Mech. 115 (5) (2018) 549–577, http://dx.doi.org/10.1002/nme. 5816.
- [73] P. Betsch, M. Schiebl, Energy-momentum-entropy consistent numerical methods for large-strain thermoelasticity relying on the GENERIC formalism, Int. J. Comput. Methods Eng. Sci. Mech. 119 (12) (2019) 1216–1244, http://dx.doi.org/10.1002/nme.6089.
- [74] A. Huerta, F. Casadei, New ALE applications in non-linear fast-transient solid dynamics, Eng. Comput. 11 (4) (1994) 317-345.
- [75] J. Bonet, A.J. Burton, A simple average nodal pressure tetrahedral element for incompressible and nearly incompressible dynamic explicit applications, Commun. Numer. Methods. Eng. 14 (1) (1998) 437–449, http://dx.doi.org/10.1002/(SICI)1099-0887(199805)14: 5<437::AID-CNM162>3.0.CO;2-W.
- [76] J.C. Simo, A framework for finite strain elastoplasticity based on maximum plastic dissipation and the multiplicative decomposition. Part II: computational aspects, Comput. Methods Appl. Mech. Engrg. 68 (1) (1988) 1–31, http://dx.doi.org/10.1016/0045-7825(88)90104-1.
- [77] A. Rodrígues-Ferran, F. Casadei, A. Huerta, ALE stress update for transient and quasistatic processes, Internat. J. Numer. Methods Engrg. 43 (1) (1998) 241–262, http://dx.doi.org/10.1002/(SICI)1097-0207(19980930)43:2<241::AID-NME389>3.0.CO;2-D.
- [78] A. Rodrígues-Ferran, A. Pérez-Foguet, A. Huerta, Arbitrary Lagrangian–Eulerian (ALE) formulation for hyperelastoplasticity, Internat. J. Numer. Methods Engrg. 53 (1) (2002) 1831–1851, http://dx.doi.org/10.1002/nme.362.
- [79] F. Armero, E. Love, An arbitrary Lagrangian–Eulerian finite element method for finite strain plasticity, Internat. J. Numer. Methods Engrg. 57 (1) (2003) 471–508, http://dx.doi.org/10.1002/nme.684.
- [80] N. Abboud, G. Scovazzi, Elastoplasticity with linear tetrahedral elements: A variational multiscale method, Internat. J. Numer. Methods Engrg. 115 (1) (2018) 913–955, http://dx.doi.org/10.1002/nme.5831.